# **Copyright Warning & Restrictions**

The copyright law of the United States (Title 17, United States Code) governs the making of photocopies or other reproductions of copyrighted material.

Under certain conditions specified in the law, libraries and archives are authorized to furnish a photocopy or other reproduction. One of these specified conditions is that the photocopy or reproduction is not to be "used for any purpose other than private study, scholarship, or research." If a, user makes a request for, or later uses, a photocopy or reproduction for purposes in excess of "fair use" that user may be liable for copyright infringement,

This institution reserves the right to refuse to accept a copying order if, in its judgment, fulfillment of the order would involve violation of copyright law.

Please Note: The author retains the copyright while the New Jersey Institute of Technology reserves the right to distribute this thesis or dissertation

Printing note: If you do not wish to print this page, then select "Pages from: first page # to: last page #" on the print dialog screen



The Van Houten library has removed some of the personal information and all signatures from the approval page and biographical sketches of theses and dissertations in order to protect the identity of NJIT graduates and faculty.

#### ABSTRACT

#### JPEG Hardware Design and a Comparative Study of JPEG and MPEG Image Compression Algorithms

by

#### Aravind Soundararajan

The complete PC-based hardware design along with the initialization and control software of the JPEG algorithm is presented in this thesis. The hardware could run in real-time to compress, decompress and display still frame images. LSI Logic JPEG chipset was used in the design. This is a dedicated chipset for the JPEG compression algorithm.

A simulation software was used for studying the performance of MPEG video compression algorithm. Various video test sequences were compressed through the codec simulation software and the resulting rate-distortion performance was calculated. These are compared with the performance of JPEG based image sequence compressor.

It is found that the MPEG-based video codec significantly outperforms the JPEGbased codec for the test sequences considered at low to medium bit rates.

## HARDWARE DESIGN FOR JPEG AND A COMPARATIVE STUDY OF JPEG AND MPEG IMAGE COMPRESSION ALGORITHMS

by

Aravind Soundararajan

A Thesis Submitted to the Faculty of New Jersey Institute of Technology in Partial Fulfillment of the Requirement for the Degree of Master of Science in Electrical Engineering

Department of Electrical and Computer Engineering

October 1993

 $\bigcirc$  $\langle$ 

#### APPROVAL PAGE

## Hardware Design for JPEG and a Comparative Study of JPEG and MPEG Algorithms

## Aravind Soundararajan

Dr.Ali N. Akansu, Thesis Advisor	
Associate Professor of Electrical Engineering, N.	JIT

Dr. Yeheskel Bar-Ness, Committee Member Distinguished Professor of Electrical Engineering, NJIT (date)

(date)

Dr.David Cooper, Committee Member GEC Marconi Electronic Systems Corporation, Totowa, NJ

(date)

## **BIOGRAPHICAL SKETCH**

Author: Aravind Soundararajan

Degree: Master of Science in Electrical Engineering.

Date: October 1993

#### Undergraduate and Graduate Education:

- Master of Science in Electrical Engineering, New Jersey Institute of Technology, Newark, NJ, 1993
- Bachelor of Science in Electronics and Communication Engineering. Regional Engineering College, Trichy, India, 1990

Major: Electrical Engineering.

#### **Presentations and Publications:**

Soundararajan Aravind, "Image Processing in Intra Venous Cerebral Angiography." IEEE Symposium, PSNA College of Engineering, Dindigul, India, January 1990 This thesis is dedicated to my parents and my brother

#### ACKNOWLEDGEMENT

I would like to thank my advisor Dr. Ali N. Akansu, for his invaluable advice, guidance and support in making this thesis a very rewarding and successful one. I also appreciate his constant questioning and recommendations which made this work and my educational goals more meaningful. I have learned a great deal from him not only academically but also in other general aspects of life.

I am highly grateful to Andria Wong and Fure Ching Jeng from Bellcore, Morristown for their cooperation in providing the MPEG simulation software and during the various stages of the project.

I also thank the committee members - Dr. Bar-Ness and Dr. Cooper who spared some of their valuable time to go through my report and give suggestions.

I am grateful to Ronald Buell, David Cooper and Mark D'amato for the help and facilities they rendered while at GEC in addition to their cooperation without which this project could never have become a reality.

Thanks and regards to my parents for their moral support and love and also to my labmates Adil, Mehmet, Ferhat and Nadir who made the period of this work a pleasant one.

## TABLE OF CONTENTS

Chapter

1 INTRODUCTION
2 JPEG STANDARD
2.1 Color Systems
2.2 The Baseline System
2.3 DCT
2.3.1 Models for DCT based systems
2.3.2 Level shift
2.4 Variable Length Coding (Huffman Coding ) 10
3 JPEG HARDWARE 12
3.1 Terms and Concepts
3.2 Typical System Application
3.3 Color and Raster-Block Converter
3.3.1 RGB Interface
3.3.2 Conversion Logic
3.3.3 Strip Memory Interface
3.3.4 YCbCr Interface
3.3.5 Internal Registers
3.3.6 Control Signals
3.4 Configuration and Control
3.4.1 Address Register
3.4.2 Gamma Correction Table
3.4.3 Configuration Registers
3.4.4 Window Control Registers
3.5 Reset Operation
3.6 Initialization Sequence

Chapter	Page
3.7 L64735 DCT Processor	35
3.7.1 Signed Forward DCT	35
3.7.2 Signed Inverse DCT	37
3.7.3 Unsigned Inverse and Forward DCT	
3.7.4 Signal Descriptions	
3.8 Configuration and Control.	40
3.9 L64735 Operation modes	40
3.10 JPEG Coder (L64745)	42
3.10.1 Signal Description	42
3.10.2 Code Interface Signals	45
3.10.3 Control Signals	46
3.11 Configuration and Control	47
3.11.1 Group ID Word	47
3.12 Hardware Description	55
3.12.1 Operation	55
4 MPEG STANDARD	56
4.1 MPEG Overview	56
4.2 Basic Concepts	58
4.3 Simplified Decoder Model	62
4.4 Preprocessing	63
4.5 Conversion from CCIR601 video	65
4.6 Motion Compensation and Estimation	67
4.7 Motion Estimation	70
4.8 Coding of Motion Vectors	72
5 RESULTS AND CONCLUSIONS	75
5.1 Hardware Implementation	75

Chapter	
5.2 Simulations	75
5.3 Conclusions and Future Directions	75
6 APPENDIX	81
7 REFERENCES	112

## LIST OF FIGURES

Figure		Page
2.1	JPEG Block Diagram	5
2.2.1	Baseline DCT Based JPEG System	6
2.3.1	Models for DCT Based Systems	9
3.1	JPEG Chipset in Encoder Application	14
3.1.1	Active Image Area.	14
3.1.2	Typical Image Correction Function	15
3.1.3	Block Processing by L64765	18
3.1.4	Data Interleaving.	19
3.1.5	Block Input Ordering	19
3.2.1	Typical System Application.	21
3.3.1	L64765 Logic Symbol	23
3.4.1	Address Register Write	28
3.4.2	Gamma Table Read and Write	29
3.4.3	Control Register.	31
3.5.1	Reset Operation.	36
3.7.1	Signed Forward DCT Configuration.	36
3.7.2	Signed Inverse DCT Configuration	38
3.7.3	Unsigned Forward DCT Configuration	38
3.7.4	Unsigned Inverse DCT Configuration	38
3.9.1	Forward DCT Configuration	44
3.9.2	Forward DCT Output Waveforms.	44
3.9.3	Inverse DCT Waveforms.	44
3.11.1	Read Initialization Sequence	48
3.11.2	Write Initialization Sequence	48
3.11.3	Configuration Group Format	49

Figure	Page
3.12.1a	Hardware Description
3.12.1b	Hardware Description
3.12.1c	Hardware Description
4.1	Overview of MPEG Coding 57
4.2.1	Macro Block Structure
4.2.2	Block Structure
4.2.3	Typical Sequence of Pictures in Bitstream Order
4.3.1	Simplified Decoder Implementation
4.4.1	Typical Sequences of Pictures in Display Order 64
4.4.2	Typical Sequence of Pictures in Bitstream Order
4.4.3	Coding and Decoding Process
4.5.1	Source Input Format with Significant Pel Area Shaded Dark
4.6.1	Group of Pictures in Display Order
5.1	JPEG Simulations
5.2	MPEG Simulations

## LIST OF TABLES

Table

3.4.1	Configuration Addresses
3.4.2	Data Formats
3.8.1	Configuration Inputs
3.10.	1 FIFOST[1:0] Status in Encoder Configuration
4.8.1	Code for Forward_f or Backward_f equal to 1,2,3
4.8.3	Modulus for Motion Vectors
5.1	JPEG Simulations
5.2	MPEG Simulations

#### Chapter 1

### INTRODUCTION

Mankind is continuously trying to transfer information at a rapid rate. The need to transfer data rapidly often necessitates that information be sent over a channel at a rate that exceeds the channel capacity. Some distortion inevitably results in such situations. In order to keep this distortion to a minimum level, we must first order the data generated by the source in accordance with it's importance at the receiver, and then either compress or discard the less significant information prior to transmission. The schemes devised to extract the significant information from the output of source and to reduce redundancy are called source coding algorithms.

With the recent advances in modern communications, signal processing and VLSI technologies the transmission and storage of real time video has become a reality. Typical Color Television or video-conferencing images have a spatial resolution of approximately 512 X 512 pixels per frame with 24 bits/pixel resolution and requires 30 frames/second to avoid any jerky motion of the picture. This is equivalent to the data rate of about 180 X  $10^6$  bits per second. Thus the channel capacity of  $10^6$  bits/second to  $10^8$  bits/second may be needed depending on the application. The transmission and storage of such a large amount of data requires data compression algorithms.

The primary application of data compression is in the transmission and storage of information.Image transmission applications span the broadcast television, remote sensing via satellites, teleconferencing, computer communications, facsimile transmission and others. The compression techniques of these applications should consider the size and complexity of the required hardware.

Video Compression methods exploit on source redundancies and the characteristics of Human Visual System (HVS). The correlation in still images and in spacio-temporal correlations in video signals are reduced. The reduction in correlation in spatial domain is called intra-frame compression, while compression in temporal domain is called inter-frame compression. Generally, the methods that achieve high compression ratios (10:1 to 50:1 for images and 50:1 to 200:1 for video) are lossy such that the reconstructed data are not identical to the original.

Lossless methods do exist, but their compression ratios are far lower, perhaps no better than 3:1. Such techniques are used only in sensitive applications such as medical images. For example, artifacts introduced by a lossy algorithm into an X-ray radiograph may suggest an incorrect interpretation and alter the diagnosis of a medical condition. Conversely, for commercial, industrial and consumer applications, lossy algorithms are preferred because they save on memory and communication bandwidth.

These lossy algorithms also generally exploit aspects of the human visual system. For example, the eye is much more receptive to fine details in the luminance (or brightness) signal than in the chrominance (or color) signals. Consequently, the luminance signal is usually sampled at a higher spatial resolution. In broadcast quality television, the digital resolution of the sampled luminance signal is 720 X 480 pixels, while for the color signals it maybe only 360 X 240 pixels. Second, the encoded (or compressed) representation of the luminance signal is assigned more bits (a higher dynamic range) than the chrominance signals.

Also, the eye is less sensitive to energy with high spatial frequency than with the low spatial frequency. Indeed, if an image on a 13 inch personal computer monitor were formed by an alternating spatial signal of black and white, the human viewer would see a uniform gray instead of the alternating checkerboard pattern. This deficiency is exploited by coding the high frequency coefficients with fewer bits and the low ones with more bits.

All these techniques add up to powerful lossy compression algorithms. In many subjective tests, reconstructed images that were encoded at 20:1 compression ratio are hard to distinguish from the original. Video data, even after compression at ratios of 10 : 1, can be decompressed with close to the analog tape quality.

There are various standard Image compression schemes. These are :

1) <u>CCITT H.261</u>: The H.261 standard of the International Consultative Committee for Telephones and Telegraphs (CCITT) specifies a method of communication for visual images and makes use of motion compensation and adaptive Discrete Cosine Transform(ADCT).

2) JPEG : The Joint Photographic Experts Group (JPEG) formed in 1986, has proposed an international standard for the compression of continuous tone (gray scale or color) images. The transform used is the Discrete Cosine Transform along with Huffman variable length codes.

3) <u>MPEG</u>: Moving Pictures Expert Group (MPEG) proposed an international standard for motion pictures which made use of motion estimation and compensation techniques leading to greater reduction in the redundancy than the JPEG. This also makes use of the DCT.

The following chapters are organized as follows. Chapter 2 explains the JPEG algorithm, Discrete Cosine Transform and Huffman coding techniques. Chapter 3 explains the Hardware designed for the JPEG algorithm. Chapter 4 deals with the MPEG algorithm. Chapter 5 explains the simulations performed and the results of the simulations. This also gives the conclusions and the future directions for research.

The objective of this thesis is to design the hardware for the real time compression and decompression of image sequences and to compare the performance of JPEG and MPEG based codecs on the same test sequence. Based on the simulations done it has been outrightly proved that MPEG based systems perform much better than the JPEG based ones for the same bits/pel and same test sequence. The tables of the Signal to Noise ratios for the two cases are shown in Chapter 5.

#### Chapter 2

#### JPEG STANDARD

JPEG stands for the Joint Photographic Experts Group Committee which is a research team formed under the auspices of the International Standards Organization (ISO). The proposed standard is a still picture coding algorithm . The scope of the algorithm is broad- it comprises a baseline lossy approach and an extended lossless approach, as well as independent functions using different coding techniques than the baseline scheme. The hardware developed in this work falls under the lossy baseline approach. The JPEG algorithm is depicted in Fig 2.1.

#### 2.1 Color Systems

The JPEG algorithm falls under the heading of transform-based image coding. A color image can be represented in different color formats. Those which are widely used today include R-G-B (the three primary colors red, green and blue) in the computer industry; Y-U-V (Y for luminance, U and V for the color difference signals Y-R and Y-B respectively) in the television industry; and the C-M-Y-K (cyan, magenta, yellow and black) in the printing industry. Within each color system, the constituent parts are called components. Thus, there are three color components in the R-G-B system and four in the C-M-Y-K system.

#### 2.2 The Baseline System

A "baseline system" as shown in Fig 2.2.1 is defined as one which guarantees that a reasonable level of function will be present in all decoders which use the DCT algorithms. This baseline system uses a constrained version of the sequential DCT algorithms. The baseline system must be present in all systems which use the DCT compression algorithms.



Fig 2.1 JPEG Block Diagram



Fig 2.2.1 Baseline DCT Based JPEG System

The baseline DCT system capabilities can be extended in a number of ways. These so called "extended systems" can use progressive modes of compression, higher precision, and an alternative coding techniques. They have a fewer constraints than the baseline system in several other important parts of the system.

A separate lossless coding algorithm is defined which uses a DPCM (Differential Pulse Code Modulation) algorithm. In this case the subsequent sample values are predicted based on the previous sample values. The baseline DCT system is not required in a lossless decoding system.

The compression achieved with these algorithms depends on the characteristics of the test images. For the color images of natural scenes which have been used for testing and development of the algorithms, recognizable images are obtained at about 0.15 bits per pixel and acceptable quality images are obtained at about 0.25 bits per pixel. At 0.75 bits per pixel, the image quality is very good, and at about 1.5 bits per pixel the images obtained are essentially indistinguishable from the original source. Lossless compression ratio is about 2:1 for these test images. All of these experiments are based on tests of 4:2:2 CCIR601 format Y,Cb,Cr color images with 16 bits per pixel.

Chapter 5 gives the results of the simulations performed with CCIR601 test images on the JPEG board and the MPEG simulation software.

#### 2.3 DCT

The term image transform usually refers to a class of unitary matrices used for representing images. One-dimensional signal can be represented by an orthogonal series of basis functions, an image can also be expanded in terms of a discrete set of basis arrays called basis images. These basis images can be generated by unitary matrices [2]. To make the transform coding practical, a given image is divided into NM/pq blocks, each of size p x q, the main storage requirements for implementing the transform are reduced by a factor of MN/pq. The computational load is reduced by a factor of

 $\log_2 MN/\log_2 pq$  for fast transform requiring  $\alpha$  Nlog<sub>2</sub> operations to transform an N X 1 vector. For 256 X 256 images divided into 8 X 8 blocks, these factors are 1024 and 2.66 respectively [4]. Although the operation count is not greatly reduced, the complexity of the hardware for implementing small size transform is reduced significantly. However, smaller block sizes yield lower compression.

The N X N forward 2D DCT is given by [2]:

$$BD(u,v) = \frac{1}{4} C(u)C(v) \sum_{y=0}^{N} \sum_{x=0}^{N} bd(x,y) \cos[\pi u(2x+1)/16] \cos[\pi v(2y+1)/16]$$

$$bd(x,y) = 1/4 \sum_{u=0}^{N} \sum_{v=0}^{N} C(u)C(v)BD(u,v)cos \left[\pi u(2x+1)/16\right]cos \left[(\pi v(2y+1)/16\right]$$

where bd(x,y) = 2-D sample value

$$BD(u,v) = 2-D DCT Coefficient$$

- u horizontal frequency index
- v vertical frequency index
- x horizontal position index
- y vertical position index

$$C(i) = \begin{cases} 1/(2)^{1/2} \text{ for } i = 0\\\\1 \text{ for } i = 1 \text{ through N} \end{cases}$$

The coefficient array of DCT, BD(u,v), is ordered with the DC component in the upper left corner, increasing horizontal "frequencies" to the right and increasing vertical frequencies going down the array.



Fig 2.3.1 Models for DCT Based Systems

#### 2.3.1 Models for DCT Based Systems

The DCT based systems can be divided into three parts, as shown in fig 2.3.1. In the encoder 8 X 8 blocks of 2D input array samples are transformed using the forward DCT (FDCT). The DCT coefficients are quantized based on their variances and visual weights. The quantizer output is mapped to a set of symbols.

In the decoder the symbols are decoded by the entropy decoder and quantized DCT coefficients values are obtained. The inverse DCT (IDCT) then transforms the 8 X 8 DCT array back into an 8 X 8 array of signal values.

The distortion between I and I' in fig 2.3.1, is governed entirely by the FDCT, quantization, dequantization and IDCT procedures, with the assumption of an error free transmission.

#### 2.3.2 Level Shift

The source image input data have an unsigned representation. When coding unsigned data with P bits of precision, the input data is level shifted by subtracting  $2^{(P-1)}$  before processing with the forward DCT. For 8 bit precision, 128 is subtracted and for 12 bit precision 2048 is subtracted. After processing with the IDCT, the same level shift must be added to the output values to obtain the unsigned representation. Overflow and underflow of the IDCT output due to the hardware limitations must be appropriately clamped.

## 2.4 Variable-Length Coding (Huffman Coding)

JPEG encoder uses variable length coding or Huffman coding. The code words are chosen to maximize the compression for the image data type to be processed. The most commonly occurring events are assigned with the shortest code words. Each symbol supplied by the model is coded using a particular code word chosen from the table. The tables used by the Huffman coding technique are constructed from information and they are included in the signalling parameters. It is, therefore, possible to create Huffman tables which are appropriate for a class images, or even specifically optimized for each individual image.

In the baseline system up to two tables may be used for DC coefficient coding and up to two tables can be used for AC coefficient coding. In the extended system, up to four DC tables and four AC tables may be used.

Tables may be inherited from previous images and from previous stages. Thus, a set of tables may be used for a sequence of images.

## Chapter 3 JPEG HARDWARE IMPLEMENTATION

We have used the LSI Logic chipset for the hardware implementation of JPEG algorithm

- . The JPEG chipset consists of three components :
- . Color and Raster to Block Converter (L64765)
- . DCT Processor (L64735)
- . JPEG Coder (L64745)

These three chips implement a complete high speed still-image or video rate compression and decompression system. Figure 3.1 shows the JPEG chipset in a typical image encoder application. In decoder the data flow direction is reversed.

In encoding applications, the L64765 Color and Raster to Block converter converts the RGB values to luminance (Y) and chrominance (Cb,Cr) components and arranges the raster image frame into 8 X 8 blocks. The L64735 DCT Processor performs a discrete cosine transform on the 8 X 8 blocks to produce DCT coefficients. The L64745 JPEG Coder chip quantizes and encodes the DCT coefficients into a compressed data stream. The high speed and pipelining capabilities of the JPEG chipset support full motion video applications. All the three components of the JPEG chipset can be used in either directions for encoding and decoding applications.

#### **3.1 Terms and Concepts**

Some important terms and concepts used in the design are enunciated here.

#### Minimum Coded Unit (MCU)

The JPEG-9 standard specifies the minimum coded unit (MCU) [5]. An MCU is the smallest portion of a source image that is coded and processed in a JPEG coder. The MCU in the L64765 Color and Raster to Block Converter is 8 pixels high by 16 pixels

wide in the fixed 2:1:1 format. Each MCU contains two data blocks of the luminance (intensity) and one data block each of the chrominance (Cb,Cr) color components.

#### Active Image area

In many applications, only a portion of the source image, the active image area is processed. The L64765 Color and Raster to Block Converter defines the active image area in terms of the four parameters : height, width, height delay, and the width delay. Figure 3.1.1 shows the definition of these parameters relative to the complete image. The height and width delays are measured in pixels. The height and width are measured in MCUs. The shaded portion of the frame is said to be blanked. Blanked portions are ignored by the L64765 and are not sent to the L64735 DCT Processor.

#### Gamma Correction

To compensate the non-linear response of recording and display equipment, it is usually necessary to apply a gamma correction to the image data. Gamma correction is a non-linear gain function . The L64765 applies the Gamma correction using a look up gamma correction table implemented in the internal RAM. Figure 3.1.2 shows a typical Gamma correction function.

#### **Color Conversion**

The RGB-to-YCbCr and the vice versa color conversions are defined in the CCIR 601 recommendations. The first set of equations shows the RGB-to-YCbCr color conversion as a set of linear equations. The second set of equations transforms YCbCr-to-RGB color components. RGB and Y values are within the interval of 16 to 235 (inclusive); Cb and Cr values are in the range of 16 to 240 (inclusive). Any computation result that exceeds these values will be clipped to the boundary to eliminate any overflow or underflow occurrences.

RGB-to-YCbCr conversion [2] :

Y = (77/256)R + (150/256)G + (29/256)BCb= (131/256)R + (110/256)G - (21/256)B + 128



Fig 3.1 JPEG Chipset in Encoder Application



Fig 3.1.1 Active Image Area



Fig 3.1.2 Typical Gamma Correction Function

Cr = (-44/256)R - (87/256)G + (131/256)B + 128YCBCr-to-RGB conversion [2] : R = Y + 1.370(Cr-128)G = Y - 0.698(Cr-128) - 0.336(Cb-128)B = Y + 1.730(Cb-128)

#### **Raster to Block Conversion**

In the encoding direction, the input data to the L64765 Color and Raster-Block converter is in raster form consisting of horizontal scan lines of upto 8192 pixels. The L64765 converts the full image raster data into data blocks of eight pixels by eight pixels in raster order, the format used by the downstream devices, the L64735 DCT Processor and the L64745 JPEG Coder. Figure 3.1.3 shows how the first block on the display screen is processed by the L64765.

In the encoding direction, the L64765 stores the eight complete scan lines in an external strip memory. When the memory is filled, the L64765 begins reading data out of the strip memory into the YCbCr buffer in 8 X 8 blocks. As each data value is read, a new value is read, a new value is written into the strip memory at the same address. This read-then-write operation reduces the amount of strip memory needed.

In the decoding direction, the L64765 reads in 8 X 8 blocks until a complete set of eight scan lines is contained in the strip memory into the image area data reformatter in raster order.

#### **Data Interleaving**

The data coming out of L64745 is initerleaved with all the components of a MCU grouped together. Fig 3.1.4 and 3.1.5 show how the L64765 processes and outputs interleaved data . The L64745 outputs four blocks of data per MCU : two blocks of luminance data, followed by one block of Cb chrominance data followed by one block of Cr chrominance data.

The number of blocks of each component in the input data stream is determined by the L64745 internal parameters  $N_0$  through  $N_3$ , which are set at the initialization. The L64745 processes  $N_0$  blocks of component 0, followed by  $N_1$  blocks of component 1, and so on. In applications using a complete JPEG chipset,  $N_0$  is set equal to 2,  $N_1$  and  $N_2$  are set equal to 1, and  $N_3$  is set to 0. Figure 3.1.6 shows the Block input ordering.

#### Quantization

Quantization categorizes data into the limited number of discrete values. Quantization is a lossy process, meaning that information is lost in the quantization process. This loss is called quantization noise.

The forward and inverse quantization functions performed by the L64745 JPEG coder are given below. The data used in the quantization is stored within the L64745 in four quantization tables. The symbols used are defined in the following the equations [5].

Forward Quantization:

Q(x(i),i) = round (x(i)/q(i))

Inverse Quantization :

 $x'(i) = Q(x(i),i)^*q(i)$ 

where :

x(i)	:	DCT coefficient created by the L64735 DCT Processor	
------	---	---	--

- x'(i) : DCT coefficient generated by the L64745. Since the quantization is a lossy process, x'(i) is not necessarily equal to x(i).
- Q(x(i),i): Quantized DCT coefficient. Q(x(i),i) is the output of the quantizer and the input to the coding modules when the L64745 is in the DCT encoding configuration. Q(x(i),i) is the output of the decoding modules and input to the inverse quantizer when the L64745 is in DCT decoding configuration.
- q(i): Quantization stepsize from the quantization table for the component being quantized.



Fig 3.1.3 Block Processing by L64765



round(x):	Rounding function. Round(x) rounds to the nearest integer. When
	x is half way between the 2 integers, round(x) rounds up for the
	positive and down for negative values of x.
i:	Index of the DCT or received DCT coefficient, i is in the range

from 0 to 63 inclusive.

#### **3.2** Typical System Application

Fig 3.2.1 gives the three components of the JPEG chipset in a typical video real-time image processing application. This system design supports bidirectional data flow for both encoding and decoding. The text following the figure describes the function of the devices in the example.

In encoding applications, the A/D converter converts analog input RGB data into 24-bit digital RGB data as the input to the L64765.

The L64765 Color and Raster to Block converter converts RGB values to luminance (Y) and chrominance(Cb,Cr) components and converts the raster image into 8 X 8 blocks in the compression direction. In the decompression direction, it converts luminance and chrominance components into RGB and translates the 8 X 8 blocks into full-screen raster signal.

The strip memory buffers eight scan lines of the raster-ordered image to assist in the raster-to-block conversion in the L64765.

The L64735 DCT Processor performs a 2D discrete cosine transform on the 8 X 8 blocks to produce DCT coefficients.

The L64745 JPEG Coder quantizes and encodes the DCT coefficients into a compressed data stream.

The Compressed Memory stores the compressed DCT coefficients. In real-time video applications, 512 Kbytes of storage can store an average of 16 compressed images of sizes 512 X 480 X 24 at a compression of 24 to 1.



Fig 3.2.1 Typical System Application
The Master Controller provides the control signals and performs the handshaking both between the host and the L64745, and also between the host and the Compressed memory and the system bus.

The Timing Generator generates the horizontal and vertical synchronization signals.

In the decoding case, the data flow is reversed, and the D/A Converter converts the digital RGB image data from the L64765 into the analog RGB data.

### **3.3** Color and Raster-Block Converter (L64765)

Fig 3.3.1 gives the block diagram of L64765, showing all the major functional blocks. The block diagram is divided into functional subgroupings as shown by the dashed lines. Each functional grouping is explained in a subsection following the figure.

# 3.3.1 RGB Interface

The bidirectional data bus PIX[23..0] transfers raster-ordered image to and from the Pixel reformatter. Pix[23..0] are configured as either inputs or outputs depending on the direction of the conversion. The RGB interface can handle the nine data formats described under "Data formats" later in this chapter.

HSYNC and VSYNC are horizontal and vertical synchronization signals which define line frequency and the refresh frequency. BUSY signal provides a mechanism for the L64765 to delay the transfer of data over the PIX bus with respect to the internal processing rate of the specified data format.

### **3.3.2** Conversion Logic

The Gamma Correction Tables store a user defined function that transforms each pixel component.



Fig 3.3.1 L64765 Logic Symbol

The L64745 has three such tables, and each table has 256 eight bit entries. These tables can be read or written as described later in "Configuration and Control". The Color Converter uses a digital matrix to perform the color conversion specified in CCIR 601 Recommendation, Mod F [2]. The color conversion matrix may be by-passed in either direction to allow systems with images already in YCbCr space to use the raster to block conversion feature. The data after color conversion must be in the correct range, a Range Limiter clamps the data to the appropriate boundary value.

The Resampling Filters implement one of the two built in transfer functions, depending on the direction of the conversion. The filters operate only on the chrominance components Cb and Cr. In the encoding direction, downsampled image data is downsampled (decimated). In the decoding direction, downsampled image data is reconstructed (interpolated).

In the encoding direction, the Multiplexer combines the Chrominance components in the data stream sent to the memory strip interface. In the decoding direction, the Demultiplexer extracts Cr and Cb data into separate data streams for conversion to RGB by the color converter.

### **3.3.3 Strip Memory Interface**

An external strip memory stores the raster data to facilitate raster-block-conversion. Since a 16 bit YCbCr pixel must be read and written every 74 nanoseconds at 13.5 MHz, the 16- bit strip memory data bus cycle time is 37 nano seconds. Static RAMs with an access time of 25 nanoseconds are typically used in the strip memory.

When the L64765 begins processing a raster-ordered image, the Strip Memory Control places the starting address 0x0000 on ADDR[15:0] and signals a write operation via MEM\_RW. The L64765 then writes eight scan lines into the strip memory on DATA[15:0], incrementing the address with each write. Once the strip memory is full, the strip memory control reads the old strip out in block order while writing the new strip into memory in Raster order. This operation is done on a pixel basis, reading a pixel out before writing a new pixel into the strip memory. The Data Register buffers read and write data.

The 16-bit address can access upto 64K pixels. This memory addressing range is more than enough to support eight 17-inch scan lines at 300 dpi (40,800 pixels). However, the system configuration need only include the amount of memory that will actually be used. The L64765 requires the minimum amount of memory as defined in the equation below, where M is the memory size in 16-bit words and width is the width of the active window in MCUs.

$$M = 128 X$$
 width

For example, an image 512 pixels wide requires a strip memory of 4K (4096) 16 bit words.

### 3.3.4 YCbCr Interface

The bidirectional data bus YCbCr[7:0] transfers converted data to and from the YCbCr Buffer. During raster-block operation, the 8-bit wide bus passes pixel data organized in 8 X 8 blocks to and from the L64735 DCT processor. An optional Y-only format is supported by forcing the Cr and Cb components to be black (0x80). This allows the YCbCr interface to operate at the same rate regardless of the format.

Three signals specify the state of the YCbCr interface. BS indicates the start of a block of 64 data samples. LBS indicates the end of last block of 64 data samples. The L64765 asserts the FRQ signal to stop the flow of data out of the YCbCr bus during encoding operations, or into the YCbCr bus during encoding operations, or into the YCbCr bus during encoding operations.

# **3.3.5 Internal Registers**

The address registers point to one of the registers in the Configuration Register File or to an address in the Gamma Correction Tables. The Configuration Register File includes five registers : Width, Height, WDelay, HDelay, and Control.

# **3.3.6 Control Signals**

The Internal Register Control Logic accepts three inputs : READ\* and WRITE\* control the direction of the access to the internal register addressed by R[1:0] and the contents of the address registers. The output enable signal OE\*, controls the output buffers for signals PIX[23:0], HSYNC, VSYNC, DATA[15:0],YCbCr[7:0],LBS and BS. (The bidirectional buffers are not shown in the block diagram). CLK is the clock for all internal registers and memory elements. The clock frequency is 27 MHz.

# **3.4 Configuration and Control**

This section describes the configuration and initialization of the L64765. It also shows functional waveforms for the read and write waveforms for the read and write operations.

### **3.4.1 Address Register**

The initialization of address register is done in the following way :

Address Register Write:

- Set R[1:0] to 00.
- Assert Write\* LOW for at least the minimum pulse width. When WRITE\* is asserted, the L64765 holds the PIX[23:16] while WRITE\* is asserted.
- Place the new value on PIX[23:16] while WRITE\* is asserted.
- The new value is stored in the address register on the rising edge of WRITE\*. Address Register Read :
- Set R[1:0] to 00.

- Assert READ\* LOW for at least the minimum pulse width. The L64765 places the Address Register value on PIX[23:16] after the falling edge and holds it past the rising edge of READ\*.

Fig 3.4.1 gives the address read and write waveforms.

# **3.4.2 Gamma Correction Tables**

The L64765 has three separate gamma correction tables, one each for the Red, Green , and Blue digital inputs. These are 256 entries for red, green, and blue. The Address register points to a gamma correction table location when R[1:0] is set to 01. The data written to a gamma address must be written in order: Red, Green, then Blue. Once the Blue value has been written, the address Register automatically increments by one.

Gamma Table Write :

To write the gamma correction tables:

- Set the Address Register to the starting address of the write as described earlier in the "Address Register".
- Set R[1:0] to 01.
- Assert Write\* for at least the minimum pulse width.
- Place the Red value on PIX[23:16] while WRITE\* is asserted.
- The value is written into the Red gamma table on the rising edge of the WRITE\*.
- Write the Green value in the same way as Red.
- Write the Blue value in the same way as Red.
- Once the Blue value is written, the address register automatically increments by one.
- Repeat the steps 3, 4, 5 to write successive values into the gamma table.

For reading the written value the same procedure is followed, but instead of WRITE\* signal READ\* is used and instead of placing the values on the PIX bus, the chip places



Fig 3.4.1 Address Register Read



Fig 3.4.2 Gamma Table Read and Write

the values itself on the PIX bus at the rising edge of READ\*. Figure 3.4.2 shows the write and read operations for Gamma Tables.

# **3.4.3 Configuration Registers**

L64765 contains five internal configuration registers. The eight-bit Control Registers set the data format of the input RGB data on the PIX bus and controls the other aspects of operation.

The Configuration registers are addressed in the same manner as the gamma correction tables. For all registers except the Control register, two write operations are required to write the full contents of the register. The least significant byte is written first to the address shown in Table 3.4.1 and the most significant byte is written to the next higher address. Only the four least significant bits of the Address Register are used to address the configuration registers.

The registers are written and read in the same manner as for gamma correction tables, with R[1:0] set to 10.

### **Control Register**

The Control Register is accessed by setting the bit 3 of Address Register to 1 and R[1:0] to 10. Figure 3.4.3 shows the content of the Control Register.

### 1) Go Control (GO)

Go enables the L64765 to begin processing the data at the next rising edge of VSYNC. After GO is set, it can only be cleared by resetting the L64765. The L64765 is reset by a write operation with R[1:0] set to 11.

### 2) Output Enable Control (OEN)

OEN and control input OE\* work together to enable and disable the state of all bidirectional signals. When either OEN is set to HIGH or OE\* is deasserted (HIGH), the

Register	# of Bytes		
Width	2		
Height	2		
WDelay	2		
HDelay	2		
Control	1		
	Register Width Height WDelay HDelay Control		

 Table 3.4.1 Configuration Addresses



Fig 3.4.3 Control Register

bidirectional signals are held in the high impedance state. For these signals to be enabled, OEN must be set to zero and OE\* set to zero.

#### 3) Code Operation Control (ENC)

ENC controls the direction of data flow, encoding or decoding. When ENC is set to one, the L64765 data flow is in the encoding direction : PIX[23:0] are the inputs and YCbCr[7:0] are the outputs.

#### 4) Raster -Block Conversion Bypass Control

RBC controls the raster-block conversion. When RBC is reset to zero, input data is assumed to be raster-ordered for the encoding direction, and block ordered for the decoding direction. The L64765 performs the appropriate conversion (raster-to-block or block-to-raster). When RBC is set to one, the conversion is by-passed; inputs are passed to outputs with no change in the data order.

#### 5) Format Control

FMT is a four bit field that controls how the L64765 interprets the PIX and YCbCr buses. Table 3.4.2 shows the meaning of the FMT field.

# **3.4.4 Window Control Registers**

The Window control registers contain the values of the four parameters WIDTH, HEIGHT, WDELAY and HDELAY. These parameters are to locate the active window size as shown in fig 3.1.1 . Each Window control register is 16 bits wide , though not all of the bits are used in each register. The bits that are not used return to zero when read. Table 3.4.1 shows the format of each register. The fields are defined below :

#### 1)WIDTH : Image Width(9bits)

The width of the active image area in MCUs (each MCU is 16 pels wide). WIDTH is in the range from 1 to 511 inclusive.

		Block Ordering Resampling Color				
FMT	PIX[23:0]	RBC=0	RBC=1	Filters	Conversion	
0	16-Bit YCb/Cr	YYCbCr	YCbYCr	No	No	
1	24-Bit CrYCb	YYCbCr	YCbYCr	Yes	No	
2	24-Bit RGB	YYCbCr	YCbYCr	Yes	Yes	
3	15-Bit RGB Targa	YYCbCr	YCbYCr	Yes	Yes	
4	16-Bit RGB i860	YYCbCr	YCbYCr	Yes	Yes	
5	16-Bit RGB XGA	YYCbCr	YCbYCr	Yes	Yes	
6	32-Bit RGB	YYCbCr	YCbYCr	Yes	Yes	
7	32-Bit BGR	YYCbCr	YCbYCr	Yes	Yes	
8	Packed-Y	ΥΥΥΥ	YYYY	No	No	
9	YCb/Cr Y-only	YYCbCr <sup>1</sup>	YCbYCr <sup>1</sup>	No	No	
10	RGB Y-only	YYCbCr <sup>1</sup>	YCbYCr <sup>1</sup>	Yes	Yes	
11-15	Reserved					

 Table 3.4.2 Data Formats

1. Cb and Cr are set to black (0x80) in these formats

#### 2)HEIGHT : Image Height (10 bits)

The height of the active image area in MCUs (each MCU is 8 pixels high). HEIGHT is in the range of 2 to 2023 inclusive.

#### **3)WDELAY : Width delay (13 bits)**

The delay in pixels from the start of the scan line to the first active pixel. The rising edge of HSYNC indicates the start of the scan line. WDELAY is in the range of 0 to 8191 inclusive.

#### 4)HDELAY : Height delay (13 bits)

The delay in scan lines from the start of the frame to the first active scan line. The rising edge of VSYNC indicates the start of the frame. HDELAY is in the range of 0 to 8191 inclusive.

#### **3.5 Reset Operation**

A write operation to the Reset Register performs a reset operation on the L64765 and the value written is ignored. A reset operation clears the Go bit in the Control register but does not otherwise affect the state of L64765. The contents of the gamma correction tables and the configuration registers are unchanged by a reset operation. A reset operation also sets the FRQ output to LOW, thus allowing the system clock to reach the L64735 and L64745. Figure 3.5.1 shows the waveforms for the reset operation.

### **3.6 Initialization Sequence**

When the L64765 is powered on, the contents of the internal registers are undefined.
Before the device is put into operation, this initialization sequence should be performed:
1) Reset the L64765 as described under "Reset". Resetting clears the Go bit.

2) Set the values of WIDTH, HEIGHT, WDELAY, and HDELAY as described under "Window Control Registers".

3) Set the values of the operating parameters in the Control Register but do not set the Go bit yet.

4) Load the values into the Gamma Correction Tables as described earlier.

5) Reset the L64765 again.

6) Reset the L64735 DCT Processor and the L64745 Coder.

7) Set the Go bit. The L64765 is now ready for operation.

# 3.7 L64735 DCT Processor

The L64735 DCT Processor had four basic functional configurations:

- Signed forward DCT
- Signed inverse DCT
- Unsigned forward DCT
- Unsigned inverse DCT

During initialization, the L64735 is programmed for one of these four functional configurations. Input signals UNSIGNED, FORWARD, ORDERZZ, and BYPASS control the functional configuration and operation of the L64735.

Each of the basic functional configurations is illustrated with a block diagram and described briefly in this section.

### **3.7.1 Signed Forward DCT**

For this configuration, Forward is HIGH and UNSIGNED is held LOW. Figure 3.7.1 shows the signed forward DCT configuration.

The L64735 receives nine-bit pixel data from the L64765 Color and Raster-Block Converter on PIX[8:0]. The L64765 organizes the pixel data in 8X8 blocks . Input signal PBS indicates the start of a pixel block. The DCT Transform computes the 12-bit DCT









coefficients for the block. The Block Buffer assembles the DCT coefficients in either raster or zig zag order, depending on the state of the input signal ORDERZZ. The 8 X 8 block appears on the DCTCOEFF[11:0] with a delay of 168 cycles. DBS indicates the start of a block of DCT coefficients.

# **3.7.2 Signed Inverse DCT**

Fig 3.7.2 shows the signed inverse DCT configuration. For this, FORWARD is held LOW and UNSIGNED is held LOW. The LBS signal produced by the L64745 is propagated to PBID and drives the L64765 LBS input. The internal organization is similar to the Signed Forward DCT, with DCT and DBID flowing in the opposite direction. The IDCT Transform reconstructs the pixel data from the DCT coefficients.

# **3.7.3 Unsigned Forward and Inverse DCT**

Figures 3.7.3 and 3.7.4 shows the unsigned forward and inverse DCT configurations respectively.

Differences from signed data configurations are:

- . PIX data is eight-bits wide (PIX 0 is not used).
- . DCTCOEFF data is 11 bits wide (DCTCOEFF:0 is not used)
- . PIX data is level shifted by the ADDER ( shown as the circle with the plus sign).

# 3.7.4 Signal Descriptions

The external Interface to the L64735 is shown in the figure

# **Pixel Interface Signals:**

# PIX[8:0] (Pixel data bus)

Pix[8:0] comprise the pixel data bus . When the L64735 is operating in the forward direction, PIX[8:0] are inputs for the 8 X 8 blocks of image data sent by the L64765. When the L64735 is operating in the inverse direction, PIX[8:0] are the









Fig 3.7.3 Unsigned Forward DCT Configuration



Fig 3.7.4 Unsigned Inverse DCT Configuration

outputs for the blocks of reconstructed data sent to the L64765. When the L64735 is configured for the unsigned data PIX:0 is not used.

### PBS (Pixel Block Start)

PBS is asserted HIGH coincident with the first data sample in each block of the pixel data on PIX[8:0]. PBS is an input for forward DCT operation and output for inverse DCT operation.

### **PBID** (Pixel Block ID)

PBID is connected to DBID through a 168-cycle delay. PBID is only sampled when PBS or DBS is asserted HIGH or an integer multiple of 64 cycles after PBS or DBS was last asserted. PBID is an input for forward operation and an output for inverse operation.

**DCT Interface Signals :** These signals comprise of the interface to the L64745 JPEG Coder.

### DCTCOEFF[11:0] (DCT Coefficient Bus)

DCTCOEFF[11:0] are outputs in the forward direction, and inputs for the compressed data in the inverse direction. When the L64735 is configured for Unsigned data, DCTCOEFF.0 is not used.

### **DBS** ( Data Block Start)

DBS is asserted HIGH coincident with the first sample data in each block of pixel data on the DCTCOEFF bus. DBS is an output for the forward operation and an input for inverse operation.

### **DBID ( DCTCOEFF Block ID)**

DBID is connected to the PBID through a 168-cycle delay. PBID is only sampled when the DBS is asserted HIGH or an integer multiple of 64 cycles after DBS was last asserted. DBID is an input for the forward operation and an input for the inverse operation.

### **3.8** Configuration and Control

Four input signals, BYPASS, FORWARD, ORDERZZ, and UNSIGNED control the function of the L64735 and the format of the DCTCOEFF and PIX buses. Table 3.8.1 summarizes the configuration and control signals. To prevent data loss configuration inputs must not be changed until at least 168 clock cycles after the last input value has been received by L64735.

# 3.9 L64745 Operation Modes

This section discusses and illustrates the following functions of L64735:

Forward DCT Operation

• Inverse DCT Operation

• By-pass Operation

# **3.9.1 Forward DCT Operation**

Fig 3.9.1 shows the input and output waveforms for the forward DCT Operation. The figure assumes the block ordering of data, ORDERZZ, is held HIGH for zig zag ordering. The ordered pair notation for the PIX and DCTCOEFF buses give the location of the data word in the 8 X 8 block in (x,y) coordinates. Output data words appear contiguously synchronized with the system clock after a delay of 168 clock cycles.

# **3.9.2 Inverse DCT Operation**

Fig 3.9.2 shows the input and output waveforms for inverse DCT Operation. The figure assumes block ordering of the input, ORDERZZ, is held HIGH for zig-zag ordering. Output data words again appear contiguously synchronized with the system clock after a delay of 168 clock cycles.

Control Signal	Signal Low	Signal HIGH
BYPASS	Activate DCT processing	Bypass DCT processing
FORWARD	Forward DCT opeartion	Inverse DCT operation
ORDERZZ	<b>Raster DCT Operation</b>	Zigzag-ordered DCT Blocks
UNSIGNED	Signed PIX data	Unsigned PIX data

 Table 3.8.1 Configuration Inputs

# **3.9.3 Bypass Operation**

Fig 3.9.3 shows the waveforms for data transfer in the forward direction with BY-PASS asserted HIGH. During Bypass operation, PBS and DBS are not used and their state is undefined

# 3.10 JPEG Coder (L64745)

This section describes the functioning of the L64745 which is the actual encoder and decoder chip.

# 3.10.1 Signal Description

The external Interface to the L64745 consists of the signals shown in the Fig 3.10.1. Each of the signal is explained below.

### DCTCOEFF[10:0]

DCTCOEFF[10:0] carries the DCT coefficients. It is an I/O bus. When the L64745 is configured as an encoder, DCTCOEFF[10:0] are inputs from DCT Processor. When L64745 is operating as a decoder these form the outputs for the reconstructed DCT coefficients.

### BS (Block Start)

BS is asserted HIGH coincident with the first data sample in each block of the DCT coefficients on DCTCOEFF[20:0]. BS is an input for encoder operation and an output for decoder operation.

### LBS (Last Block Strobe)

LBS is asserted HIGH following the last data sample of the last block in the image on DCTCOEFF[10:0]. LBS is an input for encoder configurations and an output for decoder configurations.

FIFOST[ 1:0]	State	Condition	Description
00	Ready	FL > FIFOTH	Data can be read from the FIFO
01	Not Ready	0 < FL <= FIFOTH	FIFO has data but is below threshold
10	Empty	FL =0	FIFO is empty
11	Overflow	FL > 32	FIFO Overflowed ; data has been lost

# Table 3.10.1 FIFOST[1:0] Status in Encoder Configuration



# **3.10.2** Code Interface Signals

# CODEDDAT[31:0] (Coded Data I/O Bus)

CODEDDAT[31:0] comprises of the coded data coming out of the JPEG Coder or coming to it in the case of decoder configuration. When the L64745 is configured as a 32-bit encoder, CODEDDAT[31:0] are outputs for the coded data. When the L64745 is configured as a 32 bit decoder, CODEDDAT[31:0] are inputs for the data to be decoded. When the L64745 is set for 16-bit operation, only CODEDDAT[15:0] are used.

### LCODE (Last Code I/O)

When LCODE is HIGH, CODEDDAT[31:0] carry the last 32-bit code word in the block. The signal in HIGH16 determines whether the data on CODEDDAT[15:0] is the more or less significant half word. LCODE is an output in encoder configuration and an input for decoder configuration.

### LCIF (Last Code in FIFO)

When LCIF is HIGH, the last 32-bit code word of the image is written into the FIFO output buffer. LCIF is high only for the cycle which the last code enters the FIFO buffer. LCIF is an output for the encoder configuration and an input for the decoder configuration.

### FIFOST[1:0] (FIFO Status Indicator)

FIFOST[1:0] indicate the status of the internal FIFO. The meaning of FIFOST[1:0] differs depending on the configuration of the L64745 as shown in Table 3.10.1. The Ready/Not ready threshold is user programmable during device initialization.

### NEXT (Next Data word)

When NEXT is asserted HIGH during encoding operations, the FIFO outputs the next data word. When NEXT is asserted HIGH during decoding operations, the FIFO accepts the next input data word. This functions the same way during initialization read or write process.

# 3.10.3 Control Signals

These input signals control the initialization and certain operational aspects of the L64745.

# HIGH16 (Half word Location)

HIGH16 is only meaningful when the L64745 is set for 16-bit operation. When this is asserted the CODEDDAT[15:0] is the more significant 16-bit half word of the 32-bit coded data word. When HIGH16 is deasserted, CODEDDAT[15:0] is the less significant 16-bit half word. For 32 bit operation HIGH16 must be deasserted LOW.

### **INIT** (Initialization)

When INIT is asserted (HIGH), the L64745 operating parameters can be read and written as described in "Device Initialization". When the INIT is deasserted (LOW), the L64745 functions as configured by bits STAT,LL, and ENC bits in the Configuration Group.

### **RW** (Read/Write Select)

RW is used only during device initialization. When RW is HIGH, the device parameters, code tables, and quantization tables can be read on the CODEDDAT bus. When the RW is low, the parameters and tables can be written from the CODEDDAT bus.

## **Reset (Device Reset)**

RESET resets the L64745. Resetting the L64745 prepares it to process new image frame.. To reset, RESET must be asserted (HIGH) for at least 2 clock cycles and deasserted for at least four clock cycles after device initialization. Resetting the L64745 clears the FIFO buffer but does not affect the contents of the quantization or code tables.

### CLK (System Clock)

CLK is the system clock that controls the internal state of the L64745. All input signals to the L64745 are sampled at the rising edge of CLK.

#### **OEN** (Output Enable)

When OEN is HIGH, DCTCOEFF[10:0], CODEDDAT[31:0], LBS, LCODE, and BS are enabled. When OEN is LOW, these signals are put in HIGH impedance state and the internal memory elements are put in a low power state.

# **3.11** Configuration and Control

Internal data is written to and read from the L64745 during the initialization sequence via the CODEDDAT bus. The internal data includes the functional configuration, operational parameters, quantization tables, and code tables.

Input signals INIT and RW control the start and type of initialization sequence(read or write). The initialization sequence begins when INIT is HIGH. RW is HIGH for read and LOW for write sequences.

The internal data is organized into nine groups shown in Table 3.11.1. During an initialization any group can be accessed at random. During a read sequence, the groups are accessed beginning with group 0. The input signal NEXT controls the transfer of data on the CODEDDAT bus. Typical Read, Write sequence is shown in fig 3.11.1.

### 3.11.1 Group ID Word

The group ID word consists of the word length identifier 8 bit and the group identifier GROUPID as shown in fig 3.11.2.

8-bit sets the word length for the code table data( groups 5-8). If 8 bit is set to one, code table data in this write sequence is written as 8-bit words on CODEDDAT[7:0]. Words longer than 8 bits are loaded in 8-bit bytes with the MSB first.

GROUPID is a four bit word whose value identifies the group as shown in Table 3.11.1. The table also shows the number of words in each group.



Fig 3.11.1 Read Initialization Sequence



Fig 3.11.2 Write Initialization Sequence

CODEDDAT Bit:	7	6	5	4	3	2	1	0	
Word 1	STAT	ENC	LL	NOPR	32BI1	WIDTH			
Word 2	0	0	0	0	С3	C2	C1	CO	
Word 3	Q3 Q2			C	21 QO		20		
Word 4	N1				NO				
Word 5	N3					N2			
Word 6	0	0	0		FIFOTH				

Fig 3.11.4 Configuration Group Format

### **Configuration Group**

The Configuration Group must be written as the first step in the initialization sequence.

This group consists of six 8-bit control words. The data must be written in the order of word 1 through word 6 as shown in Fig 3.11.2.

### 1) STAT (Statistics Configuration Selector)

When STAT is set to one, the L64745 is in statistics mode else it is in one of the configurations set by ENC and LL.

### 2) ENC (Coder Configuration Selector)

When ENC is set to 1, the L64745 functions as an encoder. When ENC is set to 0, the L64745 functions as a decoder.

### 3) LL (Lossless Configuration Selector)

When LL is set to 1, the L64745 functions as a lossless encoder. When ENC is set to 0, the L64745 performs DCT-based coding.

### 4) NOPR (Internal Predictor Control)

When NOPR is set to 1, the internal predictor is disabled else it is used.

# 5) 32BIT (CODEDDAT Bus Width Selector)

When this is set to one, the full 32 bits of the coded data bus,CODEDDAT[31:0] are used else only the lower 16 bits are used. When the L64745 is set to STAT mode (STAT set to one), the lower 13 bits of the bus, CODEDDAT[12:0], are used.

### 6) WIDTH (Input data Width )

WIDTH is a 3-bit field whose value sets the width of the input data word in lossless coding configuration. Bit 2 is the most significant bit. The number of bits in the input data word is WIDTH + 4. The possible value ranges from 4 to 10 bits. WIDTH set 111 is undefined and should not be used.

### 7) C3-C0 (Code Table Selectors)

C3-C0 select the code table to be used for each of the four components 3-0. When Cn is set to 1, code table 1 is used for the nth component. When Cn is set to 0, code table 0 is used.

# 8) Q3-Q0 (Quantization Table Selectors)

Q3 through Q0 are 2 bit fields that select the quantization table to be used for each of the four components 3-0. Each Qn selects one of the four quantization tables.

# 9) N3-N0 (Composite Sampling Ratio Selectors)

N3-N0 are 4 bit fields that select the composite sampling ratio to be used for each of the four components 3-0. A value of 0 indicates that the component is not used.

# 10) **FIFOTH (FIFO threshold)**

FIFOTH is a five-bit field whose value is the threshold for the Ready and Not Ready state in the FIFO status report, FIFOST[1:0]. The value of FIFOTH is in the range from 0 to 31.

# 11) 0 (Zero value)

Bits marked with 0 in fig 4.14 must be written zero, and should be ignored when read.

# 3.12 Hardware description

Figure 3.12.1a,b&c give the hardware description of the compression board. The components used in the test board are as follows:

1) 74LS666 : This is a 8-bit latch with readback facility. The data in the buffer could be readback with OERB\* signal.

2) 22V10 : This is a 10 input/output PAL.

3) 7C185A : This is a strip memory. This is a 8192 X 8 static Ram.

4) 74LS245 : These are octal Bus transceivers, designed for asynchronous communication between data buses.

5) CY4241 : This is a 64K FIFO.



Fig 3.12.1a Hardware Description



Fig 3.12.1b Hardware Description



Fig 3.12.1c Hardware Description

# 3.12.1 Operation

The operation of the circuit is explained here. The address and the data to the L64765 R to B Converter is written through U21. This controls the initialization of the chip along with the control registers U7 and U20. The PAL U8 supplies the necessary control signals. U2 and U3 form the strip memory which enables the Raster to Block, or vice versa conversion.

U9 supplies the initialization data to the Coder chip U13. The coded data from the Coder chip is stored in the FIFOs until decompression.

The R to B converter gets the RGB or YUV data and does the necessary conversion ie,0 raster to block or vice versa. DCT processor performs the DCT on these blocks of data and passes them to the Coder chip which performs Huffman coding on the DCT values obtained.

# Chapter 4 MPEG STANDARD

### 4.1 MPEG Overview

MPEG stands for Moving Pictures Experts Group Committee. This International Standard specifies the "Coding of Moving Pictures and Associated Audio for Digital Storage Media at upto 1.5 Mbps" [1]. The standard is ISO 11172 for which the acronym is MPEG.

The MPEG standard was developed in response to industry needs for an efficient way of storing and retrieving video information on digital storage media (DSM). An inexpensive medium was CD-Rom which could deliver data rates upto 1.2Mbps, and the MPEG standard was accordingly aimed at approximately at this data rate. The committee was instructed to develop a standard for data rates upto 1.5Mbps, and it was finally decided that the MPEG decoders should be able to handle all data rates upto 1.856 Mbps. This odd looking number was obtained by taking a telecommunications channel bandwidth of 1984 kbps and subtracting 128kbps as an allowance for audio. The standard allows much higher data rates to be specified, but it is expected that not all decoders will be able to handle such higher data rates.

Two other relevant international standards were being developed prior to the MPEG studies : H.261 by CCITT aimed at video-conferencing applications, and ISO 10918 by the ISO JPEG committee aimed at coding still pictures. Elements of both of these standards were incorporated into the MPEG standard, but subsequent development work by the committee resulted in coding elements found in neither.

ISO11172 specifies the bitstream in such a way that it is fairly straight forward to design a conforming decoder. Decoders may differ considerably in architectural and implementation details, but they have very few choices during the decoding process : the methods and the results of the decoding process are closely specified by the ISO 11172.



Fig 4.1 Overview of MPEG Coding
Decoders do have some freedom in methods of post processing and display, but the results of such post processing cannot be used in subsequent decoding steps.

The situation is quite different for encoders. The encoders can make many choices as they encode a picture: what quantization level to use, what block type to use, what coding pattern to use, and so on.

#### **4.2 Basic Concepts**

The MPEG standard defines a format for compressed digital video [1]. Although the MPEG standard is quite flexible, the basic algorithms have been tuned to work well at data rates of about 1 to 1.5 Mbps, at spatial resolutions of about 350 pels horizontally by 250 pels vertically, and picture rates of about 24 to 30 frames per second. The use of the word "picture" as opposed to "frame" is deliberate. MPEG codes progressively scans the images and do not recognize the concept of interlace. Television frames are interlaced, each frame is scanned twice with the scanning line offset by half a line in vertical direction. Each scan produces a field containing half the scan lines of a full frame. Each progressively scanned picture contains all the scan lines. Interlaced source video must be converted to a non-interlaced format before coding. After decoding , the decoder may optionally convert to an interlaced format before displaying the video.

The format of the coded video allows forward play and pause. Typical coding methods allow random access and fast forward. Decoders can be built which permit reverse play.

Compression of the digitized video comes from the use of several techniques : subsampling of the chroma information to match the human visual system (Human vision is not very much perceptible to chroma information as luma information), differential coding to exploit spatial redundancy, motion compensation (MC) to exploit temporal redundancy, discrete cosine transform (DCT) to match typical image statistics, uantization, variable length coding (VLC), and use of interpolated pictures.



Fig 4.2.1 Macro Block Structure


Fig 4.2.2 Block Structure (8 X 8 pixels)

The coding scheme is arranged in layers corresponding to a hierarchical structure. A picture corresponds to a single frame of motion video, or to a movie frame. Pictures are collected into groups-of-pictures (GOP) for random access purposes, and a set of groups of pictures forms a sequence which is the top coding layer. Pictures are divided into slices to give some immunity to data corruption. Slices are divided into macro blocks which are the units used for motion compensation and for changing the quantizer scale. Macroblocks consist of an array of 16 X 16 pels ( a pel is sometimes called a pixel). Pels are further divided into luma and chroma components. Each luma pel corresponds to one picture pel, but since chroma information is subsampled with the 2:1 ratio both horizontally and vertically, each chroma pel corresponds to 4 picture pels.

For convenience of coding, macro blocks are divided into six blocks of component pels : four blocks of luma, one block of Cb chroma, and one block of Cr chroma. This is shown in Fig 4.2.1.

Blocks are the basic coding units, and the DCT is applied at this block level. Each block contains 64 component pels arranged in an 8 X 8 array as shown in Fig 4.2.2.

There are four picture types : I pictures, or intra pictures, which are coded without any reference to any other pictures; P pictures, or predicted pictures, which are coded using motion compensation from some previous and a future I or P picture, and B pictures, or DC pictures, in which only the low frequency component (DC block average) is coded and which are intended only for a fast forward search mode.

A typical coding scheme contains a mix of I,P, and B pictures. A typical scheme may have an I picture about every 12 to 15 pictures, to give a reasonably fast random access, and two B pictures inserted between each pair of I or P pictures. A typical sequence of pictures, in display order, might be as shown in Fig 4.2.3.

The bitstream order, ie the order in which the pictures are transmitted, stored and retrieved, is not the display order, but rather the order which the decoder requires them to decode the bitstream. The bitstream order is shown in Fig 4.2.4.

#### I B B P B B P B B P B B I B B P B B P

Typical Sequence of pictures in display order

#### I P B B P B B P B B I B B P B B P B B

## Fig 4.2.3 Typical Sequence of Pictures in Bitstream Order



Fig 4.3.1 Simplified Decoder Implementation

#### 4.3 Simplified Decoder Model

A simplified block diagram of a possible decoder implementation is shown in Fig 4.3.1.

The input bitstream is accumulated in the Input Buffer until needed. The Variable Length Code (VLC) Decoder decodes the header of the first picture, picture 0, and determines that it is an I picture. The VLC Decoder produces indexes corresponding to the quantized DCT coefficients using the appropriate quantization step size, which is transmitted periodically. The coefficients are then transformed into pel values by the Inverse DCT transformer and stored in the Previous Picture Store and the display buffer. The picture may be displayed at the appropriate time.

The VLC Decoder decodes the head of the next picture, picture 3, and determines that it is a P picture. For each block and the Inverse Quantizer produces the actual DCT coefficients using the appropriate quantization step size.

The coefficients are then transformed into pel and added to the predicted block produced by applying the motion vectors to the stored previous picture. The resultant block is stored in the Future Picture Store and the Display Buffer. The picture cannot be displayed until B pictures 1 and 2 have been received, decoded, and displayed.

The VLC Decoder decodes the header of the next picture, picture 1, and determines that it is a B picture. For each block, the VLC decoder produces motion vectors giving the displacement from the stored previous or next pictures or both, and indexes corresponding to the quantized DCT coefficients of the difference block. These indices are then assembled for each block and the Inverse Quantizer produces the actual DCT coefficients using the appropriate quantization step size. The coefficients are then transformed into pel values and added to the predicted block produced by applying the motion vectors to the stored pictures. The resultant block is then stored pictures. The resultant block is then stored in the Display Buffer. It may be displayed at the appropriate time.

The VLC Decoder decodes the header of the next picture , picture 2, and determines that it is a B picture. It is decoded using the same method as for picture 1. After decoding picture 2, picture 0, which is in the Previous Picture Store, is no longer needed and discarded.

The VLC Decoder decodes the header of the next picture, picture 6, and determines that it is a P picture. The picture in the Future Picture Store is copied into the Previous Picture Store, and then decoding proceeds as for picture 3. Picture 6 should not be displayed until pictures 4 and 5 have been received and displayed.

The VLC decoder decodes the header of the next picture, picture 4, and determines that it is a B picture. It is decoded using the same method as for picture 1.

The VLC decoder decodes the header of the next picture, picture 5, and determines that it is a B picture. It is decoded using the same method as for picture 1.

The VLC Decoder decodes the header of the next picture, picture 9, and determines that it is a P picture. It then proceeds as for picture 6.

The VLC decoder decodes the header of the next picture, picture 8, and determines that it is a B picture. It is decoded using the same method as for picture 1.

The VLC decoder decodes the header of the next picture, picture 12, and determines that it is a I picture. It is decoded using the same method as for picture 0, and the process is repeated.

#### **4.4 Preprocessing**

The overall process of preprocessing is shown in fig 4.4.1. The source material may exist in many forms, e.g. files in CCIR 601 format. In general, it must be processed before being encoded. This section discusses some aspects of preprocessing.

For a given data rate and source material, there is an optimum picture rate and spatial resolution at which to code if the best perceived quality is desired. If the

I	В	В	Ρ	В	В	Ρ	В	В	Ρ	В	В	I	В	В	Ρ	В	В	Ρ
0	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18

## Fig 4.4.1 Typical Sequence of Pictures in Display Order

ł	Ρ	В	В	Ρ	В	В	Ρ	В	В	1	В	В	Ρ	В	В	Ρ	B	В
0	3	1	2	6	4	5	9	7	8	12	10	11	15	13	14	18	16	17

Fig 4.4.2 Typical Sequence of Pictures in Bitstream Order

resolution is too high, then too many bits will be expended on the overhead associated with each block leaving too few to code the values of each pel accurately. If the resolution is too high, then too many bits will be expanded on the overhead associated with each block leaving too few to code the values of each pel accurately. If the resolution is too high, then too many bits will be expanded on the overhead associated with each block leaving too few to code the values of each pel accurately. If the resolution is too high, then too many bits will be expanded on the overhead associated with each block leaving too few to code the values of each pel accurately. If the resolution is too low, the pel values will be rendered accurately, but high frequency coding artifacts (e.g. noise and blockiness) and the perceived resolution and sharpness of the image. This tradeoff is further complicated by the unknowns of the final viewing conditions, eg screen brightness and the distance of the viewer from the screen.

At the data rates of 1 to 1.5 Mbps, a reasonable choice for the picture rate is 20 to 30 pps, for the horizontal resolution is between 250 and 400 pels, and for the vertical resolution is between 200 and 300 pels.

#### 4.5 Conversion from CCIR601 Video to SIF

A popular source resolution is that specified by the CCIR601. This international standard for digital TV consists of component coded video Y,Cb,Cr. Y is the luma or luminance signal and gives gray scale video. Cb and Cr are two independent color signals. There are two options in the number of lines, picture rate, and the pixel aspect ratio. One option has 525 lines per frame at 60Hz and another has 625 lines per frame at 50Hz field rate. The luma pels are sampled at resolutions of 720 X 480 and 720 X 576, respectively.

These field rates and resolutions are too large for effective coding at data rates between 1 and 1.5 Mbps. One way of converting the source video rate is to use only the odd or even fields. This reduces the picture rate to 25 or 30 Hz. If the other fields is simply discarded, spatial aliasing will be introduced, and this may produce visible and objectionable artifacts. More sophisticated methods of rate conversion require significant computational power, but can perceptibly reduce aliasing effects.



Fig 4.4.3 Coding and Decoding Process



CCIR601 format is normally converted to SIF format. In this format the MPEG coder divides the luma component into macro blocks each containing a square of 16 X 16 pels. Unfortunately, the horizontal resolution, 360 is not divisible by 16. A better match is obtained by discarding the leftmost 4 pels and the rightmost 4 pels from each line. The remaining picture is called the significant pel area, and is defined as the dark shaded area in fig 4.5.1.

The conversion process is shown in table 4.5.1.

A small advantage of stripping pels from both edges is that any horizontal blanking is effectively removed from the coded picture. This may increase the coding efficiency slightly. Another slight advantage of coding at an integral multiple of 16 pels is that the overhead per pel is minimized.

#### 4.6 Motion Compensation and Estimation

P and B pictures use motion compensation to exploit temporal redundancy in the video sequence. Decoders construct a predicted block of pels from pels in a previously transmitted picture. Motion within a picture (e.g. panning) usually implies that the pels in the previous picture will be in a different position from the pels in the current block, and the displacement is given by motion vectors encoded in the bit stream . The predicted block is usually a good estimate of the current block, and it is usually more efficient to transmit the motion vector plus the difference between the predicted block and the current block, than to transmit the description of the current block by itself.

Consider the following typical group of pictures shown in fig 4.6.1. The I picture, picture 2, is decoded without requiring any motion vectors. The first P picture, number 5 is decoded using motion vectors from picture 2. This motion compensation is called forward motion of a macroblock, i.e. the motion of a 16 X 16 block of luma pels and the associated chroma component. Most macroblocks in a P picture usually use motion compensation. The vectors may be explicitly transmitted, or none may be transmitted.



Fig 4.6.1 Group of Pictures in Display Order

Table 4.8.1	Range of motion vectors
-------------	-------------------------

forward_f code	Motion vector range						
backward_f code	full_	full_pel=1					
1	-8	to 7.5	-16 to 15				
2	-16	to 15.5	-32 to 31				
3	-32	to 31.5	-64 to 63				
4	-64	to 63.5	-128 to 127				
5	-128	to 127.5	-256 to 255				
6	-256	to 255.5	-512 to 511				
7	-512	to 511.5	-1024 to 1023				

The transmitted vectors usually have a precision of half a pel. The maximum range of the vector is set by the forward\_f parameter in the picture header. Sometimes, if the motion is unusually large, the range may be doubled and the accuracy reduced to integer pels, by the full\_pel\_forward\_vector bit in the picture header.

A positive value of the horizontal or vertical component of the motion vector signifies that the prediction is formed from the pixels in the referenced picture which are spatially to the right or below the pixels predicted.

Not all the macroblocks in P picture use motion compensation. Some macro blocks, as defined by the transmitted macroblock type, may be intra, and these are reconstructed without motion compensation.

P picture number 8 uses forward motion compensation from picture 5. P picture always use forward motion compensation from the last transmitted I or P picture.

The B pictures may use motion compensation from the previous I or P picture, or from the next (in display order) I or P picture. This is equivalent to saying that they may use motion compensation from the last two transmitted I or P pictures.

Prediction is called forward if reference is made to a picture in the past, and called backward if reference is made to a picture in the future. For example, B picture 3 in fig 4.6.2 uses forward motion compensation from I picture 2, and backward motion compensation from P picture 5. B pictures may use both forward and backward motion compensation and average the result. This usage is called the interpolative motion compensation.

All three types of motion compensation are useful, and are typically used in coding B pictures. Interpolation prediction has the advantage of averaging any noise present, and is frequently used if matching information is present in both previous and future pictures. Forward or backward motion compensation may be more useful near the edges of pictures, since the motion vectors cannot reference any pels that are outside the edge of the picture, or where a foreground object is passing in front of a fixed or slow

moving background. The technique of coding with P or B pictures increases the coding efficiency. B pictures can have greater errors of reconstruction than I or P pictures to conserve coding bits, but since they are not used as the basis of motion compensation for future pictures, these errors do not accumulate.

#### 4.7 Motion Estimation

Motion compensation in a decoder is straight forward, but motion estimation in encoder, presents a formidable computational challenge.

Various methods are available in the encoder. The more computationally intensive methods tend to give better results, so there is a tradeoff to be made in the encoder :computational power, and hence the cost, vs the video quality.

Using some search strategy the encoder attempts to match the pels in a macroblock with those in a previous or future picture. The vector corresponding to the best match is reported after the search is completed.

#### 4.7.1 Block Matching Criterion

For matching, only the luma component is taken into consideration. Sufficient prediction is obtained by matching only the luma component. The chroma vectors are only half the amplitude of the luma vectors since the chroma bitmap resolution is one half that of the luma.

In seeking a match, the encoder must decide whether to use the decoded past and future pictures as the reference, or use the original past and future pictures. Note that the decoder has no choice, it must use the decoded pictures since it does not have access to the originals. Use of the decoded pictures by the encoder gives the smallest error in the motion-compensated picture, whereas use of the original pictures gives the smallest error in the motion vectors. The choice depends on whether the artifacts of increased noise,

## Table 4.8.2 Code for Forward\_f or Backward\_f equal to 1,2,3

0000	0011	001	-16
0000	0011	011	-15
0000	0011	101	-14
0000	0011	111	-13
0000	0100	001	-12
0000	0100	011	-11
0000	0100	11	-10
0000	0101	01	-9
0000	0101	11	-8
0000	0111		-7
0000	1001		-6
0000	1011		-5
0000	11		-4
0000	1		-3
0001			-2
0011			-1
011			_
1			0
010			1
0010			2
0001	0		3
0000	110		4
0000	1010		5
0000	1000		6
0000	0110		7
0000	0101	10	8
0000	0101	00	9
0000	0100	10	10
0000	0100	010	11
0000	0100	000	12
0000	0011	110	13
0000	0011	100	14
0000	0011	010	15
0000	0011	000	

or greater spurious motion are judged to be more objectional. There is usually little or no difference in quality between the two methods, however the MPEG committee chose to use the original pictures for the simulation results.

The mean square error of the difference between the motion compensated block and the current block is one of the matching criterion. The other one being the mean absolute difference between the motion-compensated block and the current block. This method has been used for the simulation model since it is easier to compute using software simulations.

#### **4.8 Coding Of Motion Vectors**

The motion vector of a macro block tends to be well correlated with the previous macroblock. For example, in a pan all the vectors would be roughly the same. Motion vectors are coded using a DPCM technique to make use of this correlation.

In P pictures the motion vectors used for DPCM, the prediction vector, is set to zero at the start of each slice and at each intra coded macro block. Note that macro blocks which are coded as predictive but which have no motion vector, also set the prediction vector to zero i.e. when there is no motion associated.

In B pictures there are two prediction vectors, forward and backward. Each vector is coded relative to the predicted vector of the same type. Both the prediction vectors are set to zero at the start of each slice and at each intra-coded macro block. Note that predictive macro blocks which have only forward vector do not affect the value of the predicted backward vector. Similarly, predictive macro blocks which have only a backward vector do not affect the value of the predicted forward vector.

The range of the vectors is set by two parameters- the full pel forward vector and full pel backward vector flags in the picture header. These two parameters determine whether the vectors are defined in half pel or integer pel units. A second parameter, forward\_f or backward\_f, defines the number of bits appended to the VLC codes in Table 4.8.1.

Advantage is taken of the fact that the range of displacement vector values is constrained. Each VLC word will represent a pair of difference values. Only one of the pair will yield a macro block vector falling within the permitted range. The range of the vector is limited to the values shown in Table 4.8.2. The values obtained by decoding the differential values must be kept within this range by adding or subtracting a modulus which depends on the f value given in Table 4.8.3.

# Table 4.8.3 Modulus for Motion Vectors

forward_f or backward_f	MODULUS	
1	32	
2	64	
3	96	
4	128	
5	160	
6	192	

# CHAPTER 5 RESULTS AND CONCLUSIONS 5.1 Hardware Implementation

The hardware designed for this project comprises of two boards- the interface to the PC and the actual Compression board. The PC interface latches on the address and the data from the address and the data bus of the PC. These are used by the Compression board to generate the control signals.

Basically the PAL 22V10 gets activated by passing proper addresses to it. This generates the control signals for the 74LS666 which generates the control signals for the Compression chips and also latches on the initializing data.

I have been able to initialize and readback the initializing parameters to the 3 LSI Logic chips. The chips supplied to us was Engineering prototypes and not the actual In-production chips. When I tried to pass the actual RGB data to the Raster to Block Converter, it was not accepting anything as the BUSY signal, which should be low while accepting data was always high. An order has been made to LSI Logic for supplying us with the In-Production chips and we are awaiting the shipment.

CCUBE PC based JPEG hardware was used for simulations and comparative study of JPEG and MPEG algorithms. For the MPEG algorithm, the simulation software supplied by Bellcore was used. CCUBE card is a PC add-on card and accept TARGA sequences as inputs. The outputs are also in the TARGA format.

TARGA format is explained here :

1) TARGA Header (18 bits)

2) Blue value (5 bits)

3) Green value (5 bits)

4) Red value (5bits).

5) Padding (1 bit)

It is actually 16 bits/pixel with RGB getting each 5 bits with 1 bit being unused.





$ \begin{array}{c} 0 & 19.45210 & 19.95040 & 21.32792 \\ 1 & 21.46676 & 21.88198 & 23.81563 \\ 2 & 12.42603 & 21.9505 & 23.75538 \\ 3 & 21.52938 & 22.13372 & 23.87972 \\ 4 & 21.52938 & 22.13372 & 23.87972 \\ 4 & 21.52938 & 21.98635 & 23.86458 \\ 5 & 22.92759 & 24.04908 & 26.71701 \\ 5 & 19.4849 & 20.00297 & 21.31095 \\ 6 & 21.26209 & 21.87390 & 23.74434 \\ 5 & 22.61657 & 23.84705 & 26.41353 \\ 7 & 21.25147 & 21.88098 & 23.64351 \\ 8 & 21.25147 & 21.88098 & 23.64351 \\ 8 & 21.25147 & 21.88098 & 23.64861 \\ 10 & 19.62441 & 20.13005 & 21.52374 \\ 61 & 22.36177 & 23.89225 & 26.39932 \\ 11 & 21.56996 & 22.19482 & 23.82687 \\ 11 & 21.56996 & 22.01942 & 23.82687 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.92612 \\ 12 & 21.56996 & 22.01948 & 23.82687 \\ 15 & 19.60450 & 20.07580 & 21.52001 \\ 16 & 21.9053 & 23.24400 & 22.58527 \\ 15 & 12.6005 & 22.12046 & 23.77218 \\ 16 & 21.9132 & 12.9142 & 23.91010 \\ 17 & 21.5005 & 22.1046 & 23.77218 \\ 20 & 19.47452 & 19.9943 & 23.73165 \\ 19 & 91827 & 20.073584 & 23.44637 \\ 19 & 91827 & 20.073584 & 23.44637 \\ 19 & 91827 & 20.073584 & 23.44623 \\ 21 & 91331 & 22.53928 & 23.77252 & 37.3555 \\ 23 & 21.57678 & 22.07592 & 23.92690 \\ 74 & 19.8714 & 20.5965 & 23.6202 \\ 74 & 19.8714 & 20.5965 & 23.6202 \\ 74 & 19.8714 & 20.5965 & 23.6202 \\ 74 & 19.8714 & 20.5965 & 23.6202 \\ 74 & 19.8714 & 20.5965 & 23.6202 \\ 74 & 19.8714 & 20.5965 & 23.6202 \\ 74 & 19.8714 & 20.5965 & 23.6202 \\ 74 & 19.8216 & 20.59867 & 23.292590 \\ 74 & 23.0328 & 23.77292 & 26.02625 \\ 76 & 19.7734 & 20.4928 & 23.46432 \\ 75 & 23.0328 & 23.77292 & 26.02625 \\ 76 & 19.7734 & 20.4928 & 23.46432 \\ 75 & 23.6376 & 22.49287 & 23.64643 \\ 23.9914 & 23.5915 & 25.4918 \\ 73 & 23.6376 & 24.5637 & 26.51293 \\ 31 & 23.6376 & 24.5637 & 26.51293 \\ 31 & 23.6376 & 24.5637 & 26.51293 \\ 31 & 23.637$	Frame	0.3b/p	0.4b/p	0.5b/p	Frame	0.3b/p	0.4b/p	0.5b/p
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	0	19.45210	19.95040	21.32792	51	22.88309	24.45287	27.02688
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1	21.46676	21.88198	23.81563	52	22.90358	24.54825	27.15417
$\begin{array}{c} 3 \ 21.52938 \ 22.13372 \ 23.87972 \\ 4 \ 21.52938 \ 21.9855 \ 23.80495 \\ 5 \ 22.6214 \ 23.9008 \ 26.71701 \\ 5 \ 19.4849 \ 20.00297 \ 21.31095 \\ 6 \ 21.22017 \ 21.88098 \ 23.68351 \\ 7 \ 21.25147 \ 21.88098 \ 23.68351 \\ 9 \ 21.25147 \ 21.88098 \ 23.68351 \\ 9 \ 22.61657 \ 23.84705 \ 26.41353 \\ 7 \ 22.61657 \ 23.84705 \ 26.40443 \\ 9 \ 19.62441 \ 20.13005 \ 21.52374 \\ 61 \ 22.36171 \ 22.6925 \ 26.39932 \\ 11 \ 21.55696 \ 22.19482 \ 23.82667 \\ 12 \ 21.55696 \ 22.01942 \ 23.82667 \\ 13 \ 21.55696 \ 22.0920 \ 23.85847 \\ 65 \ 22.0219 \ 23.67734 \ 26.09529 \\ 14 \ 21.55696 \ 22.0520 \ 23.85847 \\ 65 \ 22.03209 \ 23.34125 \ 25.85260 \\ 15 \ 19.60450 \ 20.07580 \ 21.52071 \\ 66 \ 21.90583 \ 22.4400 \ 23.68351 \\ 18 \ 21.48053 \ 21.9743 \ 23.7718 \\ 19 \ 21.5561 \ 22.00759 \ 23.185847 \\ 65 \ 20.0015 \ 20.73458 \ 23.44637 \\ 19 \ 21.55005 \ 21.1970 \ 23.385847 \\ 65 \ 20.0015 \ 20.7349 \ 23.88444 \\ 17 \ 21.50005 \ 21.1970 \ 23.77218 \\ 66 \ 20.00402 \ 20.71490 \ 23.75875 \\ 18 \ 21.48053 \ 21.94421 \ 23.77865 \\ 70 \ 19.94925 \ 20.70166 \ 23.74262 \\ 20 \ 19.47452 \ 19.99492 \ 23.28562 \\ 71 \ 19.86918 \ 20.59956 \ 23.62392 \\ 22 \ 21.3128 \ 22.01152 \ 23.81889 \\ 72 \ 19.86918 \ 20.59956 \ 23.62392 \\ 22 \ 21.3128 \ 22.01152 \ 23.81885 \\ 72 \ 19.86918 \ 20.59956 \ 23.62392 \\ 22 \ 21.3128 \ 22.01552 \ 23.97052 \ 73 \ 19.86918 \ 20.59956 \ 23.62392 \\ 22 \ 21.57678 \ 22.07592 \ 23.92650 \\ 74 \ 19.87794 \ 20.5165 \ 23.30205 \\ 25 \ 20.2035 \ 20.84904 \ 22.29214 \\ 75 \ 19.86918 \ 20.59956 \ 23.62392 \\ 22 \ 2.91331 \ 22.5336 \ 24.46432 \\ 75 \ 19.86918 \ 20.59761 \ 23.20579 \\ 23 \ 21.7737 \ 24.4613 \ 25.9918 \ 77 \ 19.86918 \ 20.4666 \ 23.30076 \\ 25 \ 2.04776 \ 23.90300 \ 25.97291 \ 81 \ 19.7704 \ 20.5165 \ 23.30205 \\ 25 \ 20.4173 \ 24.4613 \ 24.5212 \ 27.8566 \\ 21 \ 23.4773 \ 24.4613 \ 24.5627 \ 23.32179 \\ 81 \ 19.87041 \ 20.43774 \ 23.20578 \\ 31 \ 23.4773 \ 24.4613 \ 24.5652 \ 23.20379 \\ 73 \ 19.86018 \ 20.4666 \ 23.4773 \ 23.4666 \ 23.4757 \ 23.32179 \\ 22.25287 \ 24.54309 \ 26.64589 \ 21.8518 \ 21.8517 \ 20.4567 \ 23.32196 \\ 31 \ 23.4773 \$	2	21.42603	21.92505	23.75538	53	22.94779	24.07387	26.81944
$\begin{array}{llllllllllllllllllllllllllllllllllll$	3	21.52938	22.13372	23.87972	54	22.92759	24.04908	26.71701
519.484920.0029721.310555622.6416522.875426.75815621.22514721.8809823.683515822.5057723.8926526.69119821.2514721.8809823.748885922.4745023.9926426.40433919.6244120.1300521.523746122.3617123.6992526.399321121.5569622.1948223.826876322.2021323.6773426.095291321.5569622.0530223.886436422.1198623.6335326.266091421.5694522.057223.886476522.021323.6773426.095291321.5605722.1076223.816476522.001523.6773426.095291421.5500522.1204623.772186620.0040220.7149023.755751621.4805321.9441223.786556919.987220.7354423.46371921.3581721.9794323.731657019.9492520.7016623.742622019.4745219.9943921.288527119.9371920.6375123.387212123.312623.4705523.705237319.8870420.5956523.623022221.3767622.0795923.2460225.577223.256907419.87704120.5677423.3676921.5676723.3262623.464327519.88704120.5657723.35592220.3285 <td>4</td> <td>21.52938</td> <td>21.98635</td> <td>23.80458</td> <td>55</td> <td>22.66214</td> <td>23.90108</td> <td>26.43001</td>	4	21.52938	21.98635	23.80458	55	22.66214	23.90108	26.43001
621.2504921.8739023.744345722.6165721.8470526.41353721.2514721.8809823.683515822.5057721.8926526.69119821.2514721.8809823.748885922.4745021.922426.40443919.6244120.1300521.523746021.218122.1484324.600581019.6244120.1300521.523746122.3617123.692526.399321121.5644022.1687323.926126222.2162923.6773426.095291321.5546122.0264123.886136422.1198623.6375326.26091421.500522.1204623.772186621.9058323.2440025.852611519.6045020.0758021.520016720.0911520.707923.484411721.500522.1204623.772186820.0040220.7149023.755751821.4805321.9442123.786556919.9887220.7358423.446371921.3587121.3942123.286527119.9271920.6375123.37426221312.8576722.0759223.926907419.8704120.5857723.367212221.3128521.9949321.286527119.8704120.5857723.3672121312.8577822.0759223.926907419.8704120.5706523.40202221.3128523.47052 <t< td=""><td>5</td><td>19.4849</td><td>20.00297</td><td>21.31095</td><td>56</td><td>22.64165</td><td>23.87964</td><td>26.75815</td></t<>	5	19.4849	20.00297	21.31095	56	22.64165	23.87964	26.75815
7       21.25147       21.8038       358       22.50577       23.82265       26.69119         10       19.62441       20.13005       21.52374       60       21.21381       21.6483       24.60058         11       21.56996       22.16873       23.92612       62       22.21629       23.56520       26.39428         12       21.56996       22.19482       23.82687       63       22.0012       23.67734       26.09529         13       21.52636       22.0520       23.85847       65       22.03209       23.34125       25.6520         15       19.60450       20.07580       21.52001       66       21.90583       23.24400       25.6527         16       21.5005       22.12046       23.77218       68       20.00402       20.71490       23.7557         19       21.35817       21.97943       23.73165       70       19.98872       20.73584       24.4637         21       13.3778       22.01152       23.85189       71       19.92719       20.63751       23.30225         22       21.31285       23.47095       25.4918       71       19.92719       20.45857       23.0205         22       21.327878       23.47095	6	21.26209	21.87390	23.74434	57	22.61657	23.84705	26.41353
6       21.23147       21.30098       23.74888       59       22.47430       23.92824       26.40443         10       19.62441       20.13005       21.52374       60       21.21381       22.14843       24.60058         11       21.56640       22.16273       23.92622       26.39932       26.39932         12       21.56696       22.19482       23.82687       61       22.21629       23.56520       26.39932         13       21.52861       22.02041       23.88613       64       22.11986       23.67734       26.09529         14       21.5063       22.17936       23.91010       67       20.3209       23.44844       44       23.65260         15       19.60450       20.07580       21.52001       66       21.90583       23.24400       25.85277         16       21.5053       21.94421       23.78855       69       19.98872       20.73584       23.44637         19.3517       21.32748       20.0152       23.77218       68       20.0402       20.71490       23.74262         21.3125       21.93025       21.77218       20.0402       20.63751       23.38721         21.3125       21.93025       21.79722       26.02625	/	21.2014/	21.88098	23.68351	58	22.50577	23.89265	26.69119
10       10       10       12 <td< td=""><td>0</td><td>10 62441</td><td>21.88098</td><td>23.74888</td><td>59</td><td>22.47450</td><td>23.92824</td><td>26.40443</td></td<>	0	10 62441	21.88098	23.74888	59	22.47450	23.92824	26.40443
11       21.56140       22.16873       22.92612         12       21.52696       22.19482       23.82687       63       22.2012       23.65520       26.33468         12       21.52696       22.05320       23.85847       63       22.2012       23.67734       26.03552       26.33468         14       21.54896       22.05320       23.85847       65       22.03209       23.34125       25.85260         15       19.60450       20.07580       21.52001       66       21.90583       23.24400       25.85277         16       21.5005       22.12046       23.77218       68       20.0002       20.71490       23.75575         18       21.48053       21.9421       23.7855       69       19.98872       20.73584       23.44637         21.3285       21.99439       21.28852       71       19.92719       20.63751       23.38721         21.3285       21.99439       21.28852       71       19.8918       20.5965       23.62302         22       21.31285       21.93295       23.70523       73       19.87412       20.59657       23.30205         23       21.55768       20.03352       25.93164       79       19.77042       20.59	10	19.62441	20.13005	21.52374	60	21.21381	22.14043	24.60038
12       21.56996       22.19482       23.82687         13       21.52861       22.02213       23.67734       26.09529         14       1.54861       22.00213       23.67734       26.09529         15       19.60450       20.07580       21.52001       66       22.1023       23.67734       26.09529         16       21.563       22.17936       23.91010       67       20.09115       20.70979       23.48844         17       21.5057       21.97943       23.7718       68       20.0402       20.71490       23.75575         18       21.48053       21.97943       23.73165       70       19.94925       20.70166       23.74462         20       19.47452       19.9949       21.28852       71       19.92719       20.63751       23.38721         21       21.32767       22.01592       23.92690       74       19.87041       20.57041       3.6308         22       21.31285       23.77292       26.02625       78       19.77043       20.39472       3.24052         23.03228       23.77292       26.02625       78       19.77043       20.39472       3.17966         24       21.91331       2.35780       2.59164	11	21.56440	22.16873	21.025/1	61	22.301/1	23.03923	20.33332
13       21.52861       22.02641       23.88613       63       64       22.11986       23.63353       26.26609         14       21.54896       22.05320       23.85847       65       22.03209       23.34125       25.85207         16       21.563       22.17936       23.91010       67       20.09115       20.70797       23.48844         17       21.5005       22.17936       23.91010       67       20.09115       20.70797       23.48844         17       21.550       22.17936       23.91010       67       20.09115       20.70797       23.48844         17       21.5005       22.17936       23.91010       67       20.09115       20.70797       23.48844         17       21.5767       20.07592       23.85189       70       19.9872       20.73584       23.44637         21.5767       22.07592       23.92690       74       19.87041       20.57965       23.64202         22.131285       21.93825       20.70592       23.92690       74       19.87041       20.55065       23.0205         23.20385       20.20385       20.47095       25.4918       77       19.87041       20.50014       23.26608       20.17972       23.03228	12	21.56996	22,19482	23.82687	67	22.21029	23.50520	26.09529
14       21.54896       22.0520       23.8547       65       22.03209       23.34125       25.85260         15       19.60450       20.07580       21.52001       66       21.90583       23.24400       25.85270         16       21.550       22.17936       23.91010       67       20.07915       20.701490       23.75575         18       21.48053       21.94421       23.78855       69       19.98872       20.71490       23.75575         18       21.48053       21.94421       23.78855       69       19.98872       20.73584       24.44637         21       21.3218       22.01152       23.88189       72       19.88720       20.63751       23.38721         21       21.32185       21.99825       23.70523       73       19.88720       20.63751       23.38721         21       21.32185       22.07592       23.92690       74       19.87041       20.57041       23.64492         24       21.9131       22.3914       76       19.81795       20.50034       23.20598         23       23.03228       23.77922       26.0225       78       19.77136       20.40699       23.20379         23       23.41733       24.58013	13	21.52861	22.02641	23.88613	64	22.20213	23.63353	26.26609
1519.6045020.0758021.52016621.9058323.2440025.852771621.56322.1793623.910106720.0911520.7097923.488441721.500522.1204623.772186820.0040220.7149023.755751821.4805321.9442123.788556919.9887220.7358423.446371921.3581721.9794323.731657019.942520.7016623.742622019.4745219.9943921.288527119.9827120.5597723.35792321.5767822.0759223.2705237319.8870420.5996523.623022421.9133122.5339624.464327519.8243720.5216523.302052520.2038520.8490422.98147619.879520.4102123.265982723.0322823.7729226.026257819.7736820.4069923.203792823.1733423.8978025.931647919.7704320.3949723.179663023.4052723.9039025.972938119.7719920.4259823.243933123.1319324.5924626.615738419.8206820.4664323.920373223.0812624.7603926.644438619.8617620.4665223.326153523.2428224.7603926.64288519.8107620.4467323.25643623.4705227.18068 <t< td=""><td>14</td><td>21.54896</td><td>22.05320</td><td>23.85847</td><td>65</td><td>22.03209</td><td>23.34125</td><td>25.85260</td></t<>	14	21.54896	22.05320	23.85847	65	22.03209	23.34125	25.85260
16 $21.563$ $22.17936$ $23.91010$ $67$ $20.09115$ $20.70979$ $23.48644$ 17 $21.50005$ $22.12046$ $23.77218$ $68$ $20.09402$ $20.71490$ $23.75575$ 18 $21.48653$ $11.97943$ $23.73165$ $69$ $19.98872$ $20.73584$ $23.44637$ 19 $21.35817$ $21.97943$ $23.73165$ $70$ $19.94925$ $20.70166$ $23.74262$ $20.19.47452$ $19.99439$ $21.28852$ $71$ $19.92719$ $20.63751$ $23.38721$ $21.31285$ $21.93825$ $23.70523$ $73$ $19.88720$ $20.58577$ $23.35959$ $23.157678$ $22.07592$ $23.92690$ $74$ $19.87041$ $20.57041$ $23.64849$ $24.21.9131$ $22.53196$ $24.46432$ $75$ $19.82437$ $20.5255$ $20.20385$ $25.20.20385$ $20.84904$ $22.29814$ $76$ $19.877043$ $20.39497$ $23.26608$ $26$ $27.79853$ $23.47095$ $25.4918$ $77.197135$ $20.41921$ $23.20598$ $27$ $23.0527$ $23.90390$ $25.97293$ $81.19.77043$ $20.39497$ $23.17966$ $29$ $23.41713$ $24.59246$ $26.61573$ $84$ $19.82808$ $20.46463$ $23.29037$ $32$ $23.08126$ $24.58013$ $26.61828$ $85$ $19.81410$ $20.43577$ $23.28549$ $32$ $31933$ $25.05705$ $27.18068$ $90$ $19.82206$ $20.46577$ $23.321657$ $32.44079$ $22.96694$ <	15	19.60450	20.07580	21.52001	66	21,90583	23.24400	25.85277
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	16	21.563	22.17936	23,91010	67	20.09115	20.70979	23.48844
1821.4805321.9442123.788556919.9887220.7358423.446371921.3581721.9794323.731657019.9492520.7016623.742622019.4745219.9943921.288527119.9492520.6375123.87212121.3271822.0115223.851897219.8872020.5857723.359592321.5767822.0759223.926907419.8704120.5704123.648492421.9133122.5339624.464327519.8243720.5216523.302052520.2038520.8490422.298147619.8179520.5003423.266082622.7985323.4709525.49187719.7913520.4192123.205982723.0322823.7729226.026257819.7704320.3949723.179662923.4171324.3265226.295868019.1215219.3084621.905633023.4052723.033926.645898319.851720.4374623.070473223.2528724.54501326.678288519.8141020.4457723.289333523.242824.7603926.678288519.8141020.4457723.285343623.4407924.9727127.076118819.860720.4655723.359343623.7456124.9208727.075158919.8210320.4512723.285493724.6244225.15212 <td>17</td> <td>21.50005</td> <td>22.12046</td> <td>23.77218</td> <td>68</td> <td>20.00402</td> <td>20.71490</td> <td>23.75575</td>	17	21.50005	22.12046	23.77218	68	20.00402	20.71490	23.75575
1921.3581721.9794323.731657019.9492520.7016623.742622019.4745219.9943921.288527119.9271920.6375123.387122121.3271822.0115223.851897219.8872020.5857723.359592321.5767822.0759223.926907419.8704120.5704123.648492421.913122.5339624.464127519.8243720.5216523.302052520.2038520.8490422.298147619.8179520.5003423.266082622.7985323.4709525.49187719.7704320.3949723.10562723.0322823.7729226.026257819.7704320.3949723.179662923.4171324.3265226.295868019.713920.4259824.293163023.4052723.9039025.972938119.7704320.3949723.179663123.1773724.4813326.645898319.8851720.4625723.301423223.1319324.5924626.615738419.826820.466332.99373423.0812624.5801326.678288519.8141020.4357723.283933523.2428224.7603926.644388619.8264220.4665723.329343623.4710424.9214027.075158919.8617620.4665723.29543623.6303325.05705<	18	21.48053	21.94421	23.78855	69	19.98872	20.73584	23.44637
2019.4745219.9943921.288527119.9271920.6375123.387212121.3271822.0115223.851897219.8891820.5996523.623022221.3128521.9382523.705237319.8872020.5857723.302052321.5767822.0759223.926907419.8704120.5704123.648492421.9133122.5339624.464327519.8243720.5216523.302052520.2038520.8490422.298147619.8179520.503423.266082622.7985323.4709525.49187719.7704320.3949723.203792823.1733423.8978025.931647919.7704320.3949723.179663023.4052723.9039025.972938119.7719920.4259823.249363123.1773724.4813326.512938219.8107620.4374623.070473223.2528724.5430926.645898319.851720.4627923.3210423323.1319324.5924626.615738419.8208820.4646323.290373423.0812624.5801326.672288519.8617620.4657723.321803723.4407924.9214027.061778719.8604120.4857723.321803723.630325.0570727.180689019.8226620.4663723.920844021.9906722.96694 </td <td>19</td> <td>21.35817</td> <td>21.97943</td> <td>23.73165</td> <td>70</td> <td>19.94925</td> <td>20.70166</td> <td>23.74262</td>	19	21.35817	21.97943	23.73165	70	19.94925	20.70166	23.74262
2121.3271822.0115223.851897219.8891820.5996523.623022221.3128521.9382523.705237319.8870420.5965723.359592321.5767822.0759223.926907419.8704120.5704123.648492421.9133122.5339624.464327519.8243720.5216523.302052520.2038520.8490422.298147619.8179520.5003423.266082622.7985323.4709525.49187719.7913520.4192123.205982723.0322823.7729226.026257819.7736820.4069923.203792823.1733423.8978025.972938119.7719920.4259823.2479662923.4171324.3265226.295868019.1215219.3084621.905633123.1773724.4813326.512938219.8107620.4457723.310423223.2528724.5403926.645898319.8207620.4667323.290373420.812624.5801326.672288519.8141020.4457723.283933523.2842824.7603926.644438619.8362820.466323.290373424.9206727.075158719.8604120.4857723.285493923.6303325.0570527.180689019.8220820.4627223.302083923.6203225.0527027.14197<	20	19.47452	19.99439	21.28852	71	19.92719	20.63751	23.38721
22       21.31285       21.93825       23.70523       73       19.88720       20.58577       23.35959         23       21.57678       22.07592       23.92690       74       19.87041       20.57041       23.64849         24       21.91331       22.53396       24.46432       75       19.82437       20.52165       23.30205         25       20.20385       20.84904       22.29814       76       19.81795       20.50034       23.26608         26       22.79853       23.47095       25.4918       77       19.79135       20.41921       23.20598         27       23.03228       23.77922       26.02625       78       19.77043       20.39497       23.17966         29       23.41713       24.32652       26.29586       80       19.77199       20.42598       23.24936         30       23.40527       23.9030       25.97293       81       19.77199       20.42598       23.24936         312       23.17334       24.59246       26.61573       84       19.8208       20.46453       23.29037         32       23.28428       24.76039       26.64443       86       19.82617       20.46457       23.321042         36       23.44079	21	21.32718	22.01152	23.85189	72	19.88918	20.59965	23.62302
23       21.57678       22.07592       23.92690       74       19.87041       20.57041       23.64849         24       21.91331       22.53396       24.46432       75       19.82437       20.52165       23.30205         25       20.20385       20.84904       22.28814       76       19.81795       20.50034       23.26608         26       22.79853       23.47095       25.4918       77       19.79135       20.41921       23.20598         27       23.0228       23.77292       26.02625       78       19.77068       20.40699       23.20379         28       23.17134       4.32652       26.29586       80       19.12152       19.30846       21.90563         30       23.40527       23.90390       25.97293       81       19.77199       20.42598       23.24936         31       23.17137       24.48133       26.51293       82       19.81076       20.43746       23.07047         32       23.25287       24.54039       26.64589       83       19.82161       20.46453       23.29037         34       23.08126       24.58013       26.67828       85       19.81410       20.445457       23.32180         37       23.44079<	22	21.31285	21.93825	23.70523	73	19.88720	20.58577	23.35959
24       21.91331       22.53396       24.46432       75       19.82437       20.52165       23.30205         25       20.20385       20.84904       22.29814       76       19.81795       20.50034       23.2608         27       23.03228       23.77292       26.02625       78       19.77368       20.40699       23.20379         28       23.17334       23.89780       25.9164       79       19.77043       20.39497       23.17966         29       23.41713       24.32652       26.29586       80       19.12152       19.30846       21.90563         31       23.17737       24.48133       26.51293       81       19.77199       20.42598       23.24326         32       23.13193       24.59246       26.61573       84       19.83517       20.46279       23.31042         33       23.18193       24.59246       26.61573       84       19.81076       20.44577       23.28328         34       23.08126       24.58013       26.67828       85       19.81410       20.4577       23.28393         35       23.44079       24.92140       27.07515       89       19.86041       20.46577       23.32180         37       23.63033 </td <td>23</td> <td>21.57678</td> <td>22.07592</td> <td>23.92690</td> <td>  74</td> <td>19.87041</td> <td>20.57041</td> <td>23.64849</td>	23	21.57678	22.07592	23.92690	74	19.87041	20.57041	23.64849
2520.2038520.8490422.298147619.8179520.5003423.266082622.7985323.4709525.49187719.7913520.4192123.205982723.0322823.7729226.026257819.776820.4069923.203792823.1733423.8978025.931647919.7704320.3949723.179662923.4171324.3265226.295868019.1215219.3084621.905633023.4052723.9039025.972938119.7719920.4259823.249363123.1773724.4813326.615738219.8107620.4374623.070473223.2528724.5924626.615738419.8280820.4664323.290373423.0812624.5801326.678288519.8141020.4357723.283933523.242824.7603926.64438619.8362820.4663223.326153623.4710424.9214027.061778719.8604120.4857723.321803723.4407924.9727127.076118819.8617620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494223.9616425.06516 <td>24</td> <td>21.91331</td> <td>22.53396</td> <td>24.46432</td> <td>75</td> <td>19.82437</td> <td>20.52165</td> <td>23.30205</td>	24	21.91331	22.53396	24.46432	75	19.82437	20.52165	23.30205
2622.7983323.4709525.49187719.7913520.4192123.205982723.0322823.7729226.026257819.7736820.41691223.203792823.1733423.8978025.931647919.7704320.3949723.179662923.4171324.3265226.295868019.1215219.3084621.905633023.4052723.9039025.972938119.7719920.4259823.249363123.1773724.4813326.512938219.8107620.4374623.070473223.2528724.5430926.645898319.8351720.4627923.310423323.1319324.5924626.615738419.820820.4646323.290373423.0812624.5801326.678288519.8141020.4357723.283933523.2842824.7603926.644438619.8362820.4669223.326153623.4710424.9214727.0075158919.8607120.4655723.359343823.7456124.9208727.075158919.8217223.285494021.9906722.9669425.027119119.8220620.4601623.30208923.9801325.0527027.141979319.8310320.4512723.285494223.9801325.0527027.141979319.8243320.4647123.292264423.9616425.0657627.2043<	25	20.20385	20.84904	22.29814	76	19.81795	20.50034	23.26608
27       23.0326       23.7734       23.89780       25.93164       79       19.77368       20.40699       23.20379         28       23.17334       23.89780       25.93164       79       19.77343       20.39497       23.17366         29       23.41713       24.32652       26.29586       80       19.12152       19.30846       21.90563         30       23.40527       23.90390       25.97293       81       19.77199       20.42598       23.24936         31       23.17737       24.48133       26.51293       82       19.81076       20.43746       23.07047         32       23.25287       24.54309       26.64589       83       19.82517       20.46279       23.31042         33       23.13193       24.59246       26.61573       84       19.8208       20.44643       23.29037         34       23.08126       24.58013       26.67828       85       19.81410       20.43577       23.28393         35       23.24704       24.92140       27.06177       87       19.86041       20.48577       23.32180         37       23.44079       24.92172       7.07515       89       19.812162       20.46093       23.29754         40 </td <td>20</td> <td>22./9803</td> <td>23.47095</td> <td>25.4918</td> <td>77</td> <td>19.79135</td> <td>20.41921</td> <td>23.20598</td>	20	22./9803	23.47095	25.4918	77	19.79135	20.41921	23.20598
2023.4171324.3265226.295868019.1215219.3084621.905633023.4052723.9039025.972938119.7719920.4259823.249363123.1773724.4813326.512938219.8107620.4374623.070473223.2528724.5430926.645898319.8351720.4627923.310423323.1319324.5924626.615738419.8280820.4646323.290373423.0812624.5801326.678288519.8141020.4357723.228333523.2842824.7603926.644438619.8362820.4666323.320373423.4407924.9214027.061778719.8604120.4857723.312803723.4407924.9208727.075158919.8617620.4601623.302083923.6303325.0570527.180689019.8220620.4609323.297544021.9906722.9669425.027119119.8027820.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.013527.151849419.8226320.4447123.118364523.5915625.0637827.20439619.8226320.4447123.118364523.5915625.1574827.326189719.6859820.2166723.111094723.9912425.14628<	27	23.03220	23.77292	20.02020	78	19.77368	20.40699	23.203/9
2023.4052723.9039025.972938119.7719920.4259823.249363123.1773724.4813326.512938219.8107620.4259823.249363223.2528724.5430926.645898319.8351720.4627923.310423323.1319324.5924626.615738419.8280820.4646323.290373423.0812624.5801326.678288519.8141020.4357723.283933523.2842824.7603926.644438619.8362820.4569223.326153623.4710424.9214027.061778719.8604120.4857723.321803723.4407924.9727127.076118819.8660720.4655723.329343823.7456124.9208727.075158919.8617620.4601623.302083923.6303325.0570527.180689019.8220620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.0637827.20439619.8226320.4447123.118364523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.17412<	20	23.41713	23.83780	25.33104	1 80	19.77043	19 30846	21.90563
3123.1773724.4813326.512938219.8107620.4374623.070473223.2528724.5430926.645898319.8351720.4627923.310423323.1319324.5924626.615738419.8280820.4646323.290373423.0812624.5801326.678288519.8141020.4357723.283933523.2842824.7603926.644438619.8362820.4569223.326153623.4710424.9214027.061778719.8604120.4857723.359343723.4407924.9727127.075158919.8617620.4601623.302083923.6303325.0570527.180689019.8220620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.013527.151849419.8241320.4647123.129264423.9916425.0637827.20439519.8344420.4546523.293044523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.1741227.346589719.6859820.2166723.11094723.9912425.15748 <td>30</td> <td>23.40527</td> <td>23.90390</td> <td>25.97293</td> <td>81</td> <td>19.77199</td> <td>20.42598</td> <td>23.24936</td>	30	23.40527	23.90390	25.97293	81	19.77199	20.42598	23.24936
3223.2528724.5430926.645898319.8351720.4627923.310423323.1319324.5924626.615738419.8280820.4646323.290373423.0812624.5801326.678288519.8141020.4357723.283933523.2842824.7603926.644438619.8362820.4569223.326153623.4710424.9214027.061778719.8604120.4857723.359343723.4407924.9727127.075158919.8617620.4601623.302083923.6303325.0570527.180689019.8220620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8241320.4647123.292264423.9616425.0637827.204049519.8344420.4546523.293044523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.1741227.346589719.6859820.2166723.111094723.9912425.1574827.326189819.7759420.1700722.904994823.6376625.1574827.413249919.5648120.1716823.455234922.0432823.08699	31	23.17737	24.48133	26.51293	82	19.81076	20.43746	23.07047
3323.1319324.5924626.615738419.8280820.4646323.290373423.0812624.5801326.678288519.8141020.4357723.283933523.2842824.7603926.644438619.8362820.4569223.326153623.4710424.9214027.061778719.8604120.4857723.321803723.4407924.9727127.076118819.8660720.4655723.359343823.7456124.9208727.075158919.8617620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.013527.151849419.8241320.4647123.292264423.9616425.0637827.204049519.8344420.4546523.293044523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.1741227.346589719.6859820.2166723.111094723.9912425.1574827.326189819.7759420.1700722.904994823.6376625.1574827.413249919.5648120.1716823.455234922.0432823.08699<	32	23.25287	24.54309	26.64589	83	19.83517	20.46279	23.31042
3423.0812624.5801326.678288519.8141020.4357723.283933523.2842824.7603926.644438619.8362820.4569223.326153623.4710424.9214027.061778719.8604120.4857723.321803723.4407924.9727127.076118819.8660720.4655723.329343823.7456124.9208727.075158919.8617620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.013527.151849419.8241320.4647123.292264423.9616425.0637827.204049519.8344420.4546523.293044523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.1741227.346589719.6859820.2166723.111094723.9912425.1574827.413249919.5648120.1716823.455234922.0432823.0869925.341585023.0202224.4359527.28132	33	23.13193	24.59246	26.61573	84	19.82808	20.46463	23.29037
3523.2842824.7603926.644438619.8362820.4569223.326153623.4710424.9214027.061778719.8604120.4857723.321803723.4407924.9727127.076118819.8660720.4655723.329343823.7456124.9208727.075158919.8617620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.013527.151849419.8241320.4647123.292264423.9616425.0651627.204049519.8344420.4546523.293044523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.1741227.346589719.6859820.2166723.111094723.9912425.1574827.413249919.5648120.1700722.904994823.6376625.1574827.413249919.5648120.1716823.455234922.0432823.0869925.341585023.0202224.4359527.28132	34	23.08126	24.58013	26.67828	85	19.81410	20.43577	23.28393
3623.4710424.9214027.061778719.8604120.4857723.321803723.4407924.9727127.076118819.8660720.4655723.329343823.7456124.9208727.075158919.8617620.4601623.302083923.6303325.0570527.180689019.8220620.4609323.297544021.9906722.9669425.027119119.8027820.4422723.078704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.013527.151849419.8241320.4647123.292264423.9616425.0651627.204049519.8344420.4546523.293044523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.1741227.346589719.6859820.2166723.111094723.9912425.1462827.326189819.7759420.1700722.904994823.6376625.1574827.413249919.5648120.1716823.455234922.0432823.0869925.341585023.0202224.4359527.28132	35	23.28428	24.76039	26.64443	86	19.83628	20.45692	23.32615
37       23.44079       24.97271       27.07511       88       19.86607       20.46057       23.35934         38       23.74561       24.92087       27.07515       89       19.86176       20.46016       23.30208         39       23.63033       25.05705       27.18068       90       19.82206       20.46093       23.29754         40       21.99067       22.96694       25.02711       91       19.80278       20.44227       23.07870         41       23.62442       25.15212       27.28884       92       19.83103       20.45127       23.28549         42       23.98013       25.05270       27.14197       93       19.83103       20.45127       23.28549         43       24.02511       25.0135       27.15184       94       19.82413       20.46471       23.29226         44       23.96164       25.06516       27.20404       95       19.83444       20.45465       23.29304         45       23.59156       25.06378       27.2043       96       19.82263       20.44471       23.11836         46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124<	36	23.47104	24.92140	27.06177	87	19.86041	20.48577	23.32180
38       23.74361       24.92087       27.18068       99       19.86176       20.46016       23.30208         39       23.63033       25.05705       27.18068       90       19.86276       20.46093       23.29754         40       21.99067       22.96694       25.02711       91       19.80278       20.44227       23.07870         41       23.62442       25.15212       27.28884       92       19.83103       20.45127       23.28549         42       23.98013       25.05270       27.14197       93       19.83103       20.45127       23.28549         43       24.02511       25.0135       27.15184       94       19.82413       20.46471       23.29226         44       23.96164       25.06516       27.20404       95       19.83444       20.45465       23.29304         45       23.59156       25.06378       27.2043       96       19.82263       20.44471       23.11836         46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124       25.14628       27.32618       98       19.77594       20.17007       22.90499         48       23.63766<	37	23.440/9	24.9/2/1	27.07611	88	19.8000/	20.46557	23.35934
3923.6303323.6370327.180669019.8220620.4403323.237374021.9906722.9669425.027119119.8027820.4422723.08704123.6244225.1521227.288849219.8310320.4512723.285494223.9801325.0527027.141979319.8310320.4512723.285494324.0251125.013527.151849419.8241320.4647123.292264423.9616425.0651627.204049519.8344420.4546523.293044523.5915625.0637827.20439619.8226320.4447123.118364624.0427525.1741227.346589719.6859820.2166723.111094723.9912425.1462827.326189819.7759420.1700722.904994823.6376625.1574827.413249919.5648120.1716823.455234922.0432823.0869925.341589719.6648120.1716823.455235023.0202224.4359527.281329119.5648120.1716823.45523	38	23./4001	24.92007	27.07515	89	19.001/0	20.40010	23.30200
41       23.62442       25.15212       27.28884       92       19.83103       20.45127       23.28549         42       23.98013       25.05270       27.14197       93       19.83103       20.45127       23.28549         43       24.02511       25.0135       27.15184       94       19.82413       20.45127       23.28549         43       24.02511       25.0135       27.15184       94       19.82413       20.46471       23.29226         44       23.96164       25.06516       27.20404       95       19.83444       20.45465       23.29304         45       23.59156       25.06378       27.2043       96       19.82263       20.44471       23.11836         46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124       25.14628       27.32618       98       19.77594       20.17007       22.90499         48       23.63766       25.15748       27.41324       99       19.56481       20.17168       23.45523         49       22.04328       23.08699       25.34158       50       23.02022       24.43595       27.28132	39	23.03033	25.05705	27.10000	90	10 90270	20.40093	23.29734
41       23.02442       23.13212       27.128034       92       19.03103       20.45127       23.28549         42       23.98013       25.05270       27.14197       93       19.83103       20.45127       23.28549         43       24.02511       25.0135       27.15184       94       19.82413       20.46471       23.29226         44       23.96164       25.06516       27.20404       95       19.83444       20.45465       23.29304         45       23.59156       25.06378       27.2043       96       19.82263       20.44471       23.11836         46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124       25.14628       27.32618       98       19.77594       20.17007       22.90499         48       23.63766       25.15748       27.41324       99       19.56481       20.17168       23.45523         49       22.04328       23.08699       25.34158       50       23.02022       24.43595       27.28132	40	21.55007	22.30034	23.02/11	01	10 002/0	20.44227	23.07870
43       24.02511       25.06376       27.14197       94       19.83103       20.46471       23.2034         43       24.02511       25.06378       27.15184       94       19.82413       20.46471       23.2926         44       23.96164       25.06516       27.20404       95       19.83444       20.45465       23.29304         45       23.59156       25.06378       27.2043       96       19.82263       20.44471       23.11836         46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124       25.14628       27.32618       98       19.77594       20.17007       22.90499         48       23.63766       25.15748       27.41324       99       19.56481       20.17168       23.45523         49       22.04328       23.08699       25.34158       50       23.02022       24.43595       27.28132	41	23.02442	25.15212	27.20004	92	10 02103	20.45127	23.20319
44       23.96164       25.06516       27.20404       95       19.83444       20.45465       23.29304         45       23.59156       25.06378       27.2043       96       19.82465       20.44471       23.11836         46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124       25.14628       27.32618       98       19.77594       20.17007       22.90499         48       23.63766       25.15748       27.41324       99       19.56481       20.17168       23.45523         49       22.04328       23.08699       25.34158       50       23.02022       24.43595       27.28132	12	23.30013	25.05270	27.14197		10 02/13	20.45127	23.20275
45       23.59156       25.06378       27.2043       96       19.82263       20.44471       23.11836         46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124       25.14628       27.32618       98       19.77594       20.17007       22.90499         48       23.63766       25.15748       27.41324       99       19.56481       20.17168       23.45523         49       22.04328       23.08699       25.34158       50       23.02022       24.43595       27.28132	43	23 96164	25 06516	27 20404		19.83444	20.45465	23.29304
46       24.04275       25.17412       27.34658       97       19.68598       20.21667       23.11109         47       23.99124       25.14628       27.32618       98       19.77594       20.17007       22.90499         48       23.63766       25.15748       27.41324       99       19.56481       20.17168       23.45523         49       22.04328       23.08699       25.34158       50       23.02022       24.43595       27.28132	45	23.59156	25.06378	27.2043	94	19.82263	20.44471	23.11836
4723.9912425.1462827.326189819.7759420.1700722.904994823.6376625.1574827.413249919.5648120.1716823.455234922.0432823.0869925.341585023.0202224.4359527.28132	46	24.04275	25,17412	27.34658	97	19.68598	20.21667	23.11109
48       23.63766       25.15748       27.41324       99       19.56481       20.17168       23.45523         49       22.04328       23.08699       25.34158       99       19.56481       20.17168       23.45523         50       23.02022       24.43595       27.28132       99       19.56481       20.17168       23.45523	47	23.99124	25.14628	27.32618	98	3 19.77594	20.17007	22.90499
49 22.04328 23.08699 25.34158 50 23.02022 24.43595 27.28132	48	23.63766	25.15748	27.41324	99	19.56481	20.17168	23.45523
50 23.02022 24.43595 27.28132	49	22.04328	23.08699	25.34158	1			
•	50	23.02022	24.43595	27.28132	1			

Frame	Frame	b/p 0.	2b/p 0.	4b/p 0.5	Frame	b/p 0.	2b/p 0.	.4b/p 0.5
1	l	20.99	23.96	26.69	51	22.14	25.93	3 27.99
2	2	21.10	24.17	25.00	52	26.38	30.15	5 31.18
3	3	21.93	24.16	25.05	53	23.69	26.99	28.08
4	4	20.69	24.75	26.38	54	23.83	26.64	27.96
5	5	20.63	26.03	26.80	55	26.79	29.25	5 30.73
6	6	22.22	26.05	22.77	56	24.09	26.53	3 28.42
7	7	22.03	26.97	28.03	57	24.25	26.99	28.80
8	8	22.63	26.62	27.03	58	26.34	28.87	7 33.14
9	9	22.72	26.52	27.09	59	22.73	24.75	5 26.44
10	10	23.25	27.08	27.53	60	22.90	25.09	9 26.77
11	11	23.39	26.29	22.72	61	23.68	26.20	30.64
12	12	23.56	26.64	26.71	62	23.30	26.14	26.74
13	11	23.93	27.97	28.16	63	23.34	26.57	7 26.79
14	15	24.12	25.58	26.37	64	23.65	29.73	3 30.46
15	15	20.00	21.1/	23.81	65	23.38	20.44	2 27.30
17	17	24.00	23.40	27.04	60	24.20	20.22	7 20 70
19	19	23.63	24.10	25.19		20.30	20.47	1 23.13
19	19	24 20	24.10	22.33	60	22.24	21.40	22.75
20	20	21.11	21.44	20.33	70	22.00	23.23	24.00
21	21	22.64	23.93	22.74	71	23.22	26.09	28.31
22	22	22.62	24.02	27.74	72	23.30	25.89	27.96
23	23	23.00	24.01	25.30	73	24.00	26.33	28.75
24	24	23.40	25.03	20.55	74	23.91	28.23	29.55
25	25	23.85	24.73	25.25	75	24.15	27.81	29.09
26	26	24.93	24.87	22.32	76	24.57	28.46	5 27.66
27	27	24.99	27.46	25.87	77	24.57	27.87	7 29.99
28	28	25.43	25.89	29.14	78	24.72	28.03	3 29.87
29	29	25.56	26.36	27.36	79	20.60	25.56	5 30.44
30	30	25.06	28.16	27.60	80	21.57	25.58	3 26.20
31	31	25.06	26.63	27.37	81	20.20	28.09	24.99
32	32	25.10	26.52	31.21	82	20.41	24.00	29.41
33	33	26.34	28.71	29.29	83	20.90	28.44	30.91
34	34	27.08	27.69	33.14	84	20.03	29.39	9 32.91
35	35	26.20	28.39	28.76	85	20.02	29.33	32.54
36	36	26.13	30.72	29.15	86	21.56	29.22	2 34.14
37	37	27.77	24.71	32.44	87	21.64	30.26	5 23.81
38	38	26.70	28.72	29.21	88	21.75	30.05	29.79
39	39	26.77	28.39	29.17	89	23.34	29.57	29.39
40	40	28.03	28.15	31.49	90	23.34	30.38	29.85
41	41	27.03	28.38	25.57	91	23.49	30.64	1 33.05
42	42	23.30	30.66	29.01	92	23.80	30.15	7 32.81
43	43	20.19	20.00	32.84	93	23.84	25.74	1 24 21
44	44	20.39	20.30	29.25	94	22.33	23./*	1 22 00
4 D A E	40	20.40	20.20	23.33	30	23.21	21.13	7 34 20
40 17	40	26 02	24.33	32.JU 30 7F	50	23.23	27.37	2 30 51
	~~/ / 0	20.03	20.3/	29.79	00	21 01	20.00	, 20.9T
		20.02	27.41	27.52	00	21.91	20.24	22.26
49 50	49	20.07	34 10	24+10 25 21	100	21./0	21.0	1 27 50
Frame	Frame	h/n 0 1	27.17 h/n 0 /	1b/n 0 5	Frame	20.23	2h/n 0	4h/n 0 5

#### **5.2 Simulations**

Simulations were carried out with standard test sequence (table tennis) on the existing CCUBE JPEG hardware and Optibase MPEG software. The resulting SNR for various compression ratios were calculated. The output filesize was varied in the JPEG hardware and hence the bits/pel was varied. The input files are in Targa format which is explained in the next section. The output files are also created in Targa format. The resulting SNR between the input and output Targa files were calculated and tabulated. Table 5.1 gives the SNR for JPEG simulations for 0.2,0.3,0.4 bits/pel. MPEG simulation software creates output SIF files. These were converted into Targa files and the resulting SNR between the input and output Targa files were calculated. Table 5.2 gives the results obtained for the MPEG simulation.

Fig 5.3 gives the plot for frame index vs SNR for JPEG and fig 5.4 gives the plot for MPEG.

#### **5.3 Conclusions and Future Directions**

The results of the video codec simulation clearly indicate that for the same bit rate and bits/pel MPEG gives a much higher SNR than JPEG. However for low bit rates the performance of JPEG is commendable and much less complex than MPEG.

In the area of lossy compression the most obvious additional capability to add to JPEG is adaptive quantization. For a given image quality it has been noted that compression could be improved by as much as 30% by adaptive quantization.

Adaptive quantization is however not a cost free option, because the determination of the quantization values that are used with a given DCT block can be a computational burden, and provision for multiple quantization tables or scaling of quantization table values in the encoder and decoder must be provided. The methods adopted to select the actual quantization table to be used for a particular set of image could be time consuming and the hardware could become more complex. Although the JPEG committee's decision not to include adaptive quantization can be defended

on the basis of simplicity, there is no question that adaptive quantization would significantly improve image quality at a given bit rate.

The most obvious way of introducing adaptive quantization would be to encode a binary decision at the beginning of each MCU to tell the decoder when new quantization tables (or modification of tables) are needed. The arithmetic coder could code this as a single binary decision, perhaps conditioned on the preceding adaptive quantization decision. For best coding efficiency, the Huffman coder would probably need to run-length encode the sequence of adaptive quantization decisions. Alternatively, a "table switch " code could be defined for selecting from a predefined tables.

MPEG encoder is highly complex. As of now MPEG decoders are floating in the market with few encoders. Work is yet to be done to simplify MPEG encoder. The most difficult part in an encoder is the motion estimation. The search methods adopted by the encoder i.e, the logarithmic search and pixel recursive search are time consuming when it comes to hardware implementation. Work has to be done in this area for simplifying the hardware for this purpose.

## APPENDIX

The appendix contains a listing of the software used for initializing and controlling the LSI Logic chips and the conversion between YCbCr and RGB color space coordinates.

/\* This program initializes the L64765 \*/

#include<stdio.h>
#include<conio.h>
#include<stdlib.h>
#include<dos.h>
#include "init.h"
#define BASE 0x2a0
#define ctrlw 0x2002
#define dataw 0x2004
#define datarb 0x2003
#define initw 0x2008
#define rw 0x0000
#define cw 0x0006
#define gw 0x0002
#define bw 0x0004

#define HSYNC 0x08 #define VSYNC 0x02 #define CLK 0x04

unsigned data[48];

```
void
init_parameters(void)
{
 int i;
 for(i=0;i<15;i++)
 {
   data[i] = 0x39;
 }
for(i=15;i<48;i++)
 {
   data[i] = 0x93;
 }
}
void
write_control_rgb(unsigned add)
{
   outport(BASE + cw,add);
}
```

```
void
write_data_rgb(unsigned dat)
{
 outport(BASE + rw,dat):
}
void
write_control_jpeg(unsigned add) /* Writes the control data to 74LS666 */
{
  outport(BASE + ctrlw,add);
}
void
write_data_jpeg(unsigned dat) /* Writes the data */
{
 outport(BASE + dataw,dat);
}
void
reset(void)
ł
 write_control_jpeg(0xff7f); /* R[1:0] is set to 11 for reset */
 write_control_jpeg(0xff5f); /* W* is made low */
 write_control_jpeg(0xff7f);
}
void
write_window(void)
Ł
/* Just feed in the first 4 bytes of information and leave the
control register until the end */
/* The control data for the write cycle for a window write is
              R[1:0] is 00 and write* is high
  0xf07f
              Write* is high
  0xf05f
  0xf07f
  Control bits 11 thro' 8 set the R[1:0]
*/
int i:
 write_control_jpeg(0xf07f); /* R[1:0] is made 00 for address */
 write_control_jpeg(0xf07f);
 write_control_jpeg(0xe05f);
 write_data_jpeg(0x00);
 write_control_jpeg(0xe07f);
 write_control_jpeg(0xf07f);
 write_control_jpeg(0xf17f); /* R[1:0] is made 10 for config write */
  for(i=0;i<8;i++)
```

```
{
 write_control_jpeg(0xe15f);
 printf("DATA[%d]=%x\n",i,dat[i]);
 write_data_jpeg(dat[i]);
 write_control_jpeg(0xe17f);
 write_control_jpeg(0xf17f);
 write_control_jpeg(0xf15f);
 }
}
unsigned
write_readback_data(void)
{ unsigned ret;
 ret=inport(BASE+initw); /* Reads the readback value */
 return(ret);
}
void
readback_jpeg(void) /* This reads back from the L64765
                                                                              */
/* Control data for a read cycle is
 0xf07f
            Read* is low
 0xf03f
 0xf07f
*/
unsigned ret;
 int i;
 for(i=0;i<8;i++)
  Ł
  write_control_jpeg(0xf17f);
  write_control_jpeg(0xf13f);
  ret=write_readback_data()&0x00ff;
  printf("Read value is %02x\n",ret);
  write_control_jpeg(0xf17f);
  }
}
void
readback_gamma(void) /* This reads back the gamma table*/
{ unsigned ret,add;
  int i;
  write_control_jpeg(0xf07f);
  printf("The address to be read back\n");
scanf("%x",&add);
  write_control_jpeg(0xe05f);
  write_data_jpeg(add);
  for(i=0;i<10;i++)
  {
```

```
write_control_jpeg(0xf27f);
 write_control_jpeg(0xf23f);
 ret=write_readback_data()&0x00ff;
 printf("Read value is %x Wrote %x\n",ret,gamma[i]);
 write_control_jpeg(0xf27f);
 }
}
void
write_gamma(void)
{
 int i, j;
 unsigned temp1,temp2;
 FILE *fgamma;
 if((fgamma=fopen("gammae.tab","rb"))==NULL){
 printf(" Unable to open the gammae.tab file\n");
 exit(0);
 }
 write_control_jpeg(0xf07f);
 write_control_jpeg(0xe05f);
 write_data_jpeg(0x00);
 write_control_jpeg(0xe27f);
 write_control_jpeg(0xf27f);
 write_control_jpeg(0xe25f);
 for(i=0;i<256;i++)
 ł
  fscanf(fgamma,"%u %u %u\n ",&gamma[i],&temp1, &temp2);
  for(j=0;j<3;j++)
  Ł
  write_data_jpeg(gamma[i]) ;
  write_control_jpeg(0xe27f);
  write_control_jpeg(0xf25f);
  write_control_jpeg(0xe25f);
  }
  }
}
void
main(void)
{
 int i,enc_dec=1;
 unsigned add, ret;
 clrscr();
 if(enc\_dec==1)
 ł
 init_parameters();
```

```
reset();
write_window();
printf(" What is the address to be read back\n");
scanf("%x",&add);
write_control_jpeg(0xf07f);
write_control_jpeg(0xe07f);
write_control_jpeg(0xe05f);
write_data_jpeg(add);
readback_jpeg();
 printf("Do you want to write gamma\n");
  if(getch()=='y')
   {
write_gamma();
readback_gamma();
ł
printf(" Shall I initialise the control register\n");
if(getch()=='y')
       write_control_jpeg(0xf07f);
{
   write_control_jpeg(0xe07f);
   write_control_jpeg(0xe05f);
   write_data_jpeg(0x08);
   write_control_jpeg(0xe07f);
   write_control_jpeg(0xf17f);
   write_control_jpeg(0xe15f);
   write_data_jpeg(0x62);
   write_control_jpeg(0xe07f);
   write_control_jpeg(0xf07f);
}
   write_control_jpeg(0xf07f); /* For reading back the control value */
   write_control_jpeg(0xe07f);
   write_control_jpeg(0xe05f);
   write_data_jpeg(0x08);
   write_control_jpeg(0xf17f); /* For reading back the control value fed in */
   write_control_jpeg(0xf13f);
   ret=write_readback_data()&0x00ff;
   printf("control value is %02x\n",ret);
   write_control_jpeg(0xf17f);
```

/\* This program initializes the L64745 \*/

#include <stdio.h> #include <stdlib.h> #include <conio.h> #include <dos.h> #include "init.h" #define ctrlw 0x2002 #define BASE 0x2a0 #define rw 0x0000

}

```
#define cw 0x0006
#define rdback45 0x2007
#define write45 0x2006
#define reset45 0x2000
int check = 1;
unsigned dat[]= \{0x44,0x0f,0xe4,0x00,0x00,0x01\};
void
init_parameters(void)
ł
 unsigned dc_val[48],ac_val[400];
 int i;
 for(i=0;i<48;i++)
 ł
   dc_val[i] = 0x44;
  J
 for(i=0;i<356;i++)
 {
   ac_val[i]=0x32;
  }
}
void
clock(void)
ł
 outport(BASE + cw, 0x10);
 outport(BASE + cw,0x14);
 outport(BASE + cw,0x14);
 outport(BASE + cw,0x14);
 outport(BASE + cw, 0x14);
 outport(BASE + cw, 0x10);
}
void
write_ctrl_jpeg(unsigned dat)
ł
 outport(BASE + ctrlw,dat);
}
void
write_data_coder(unsigned dat)
{
 outport(BASE + write45,dat);
}
void
quant_table_write(unsigned group_id, int num) / * Writes
                                                                   the quantization
tables to the L64745 */
ł
```

```
int i, j, check;
 FILE *fquant;
if(num <= 3){
        printf(" Do you want to initialize the quant table %d ?\n",num);
        if(getch()=='y') {
          write_ctrl_jpeg(0xf000); /* OEN is tied low at this stage */
           write_ctrl_jpeg(0xfc18);
            write_ctrl_jpeg(0xbc18); /* '666 is enabled,OEN is enabled and write mode
*/
             write_data_coder(group_id);
               clock();
        printf(" Loading the quantization table %d\n",num);
        write_ctrl_jpeg(0xf000); /* OEN is tied low at this stage */
          write_ctrl_jpeg(0xfc18);
           write_ctrl_jpeg(0xbc18); /* '666 is enabled,OEN is enabled and write mode
*/
            write_data_coder(group_id);
            clock();
if(group_id==0x11)
    for(i=0;i<64;i++)
      write_ctrl_jpeg(0xbf18);
      write data coder(defaultYQ[i]);
      clock();
      }
   }
 else
         for(i=0;i<64;i++)
       ł
      write_ctrl_jpeg(0xbf18);
      write_data_coder(defaultCQ[i]);
      clock();
       }
   }
  group_id++;
  num += 1:
  guant_table_write(group_id,num);
 };
}
void
dc_table(unsigned group_id,int num) /* Writes the DC table */
 {
  int i=64;
  unsigned temp;
  FILE * fdc tab;
  if(num <=1)
  {
```

```
printf(" Do you want to initialize the dc table %d?Press y or n\n",num);
  if(getch()=='y')
  printf(" Loading the dc table %d\n",num);
  if (num==0)
  fdc_tab=fopen("jpegdcey.tab","rb");
  else fdc_tab=fopen("jpegdcec.tab","rb");
 write_ctrl_jpeg(0xf000); /* OEN is tied low at this stage */
write_ctrl_jpeg(0xfc18);
write_ctrl_jpeg(0xbc18); /* '666 is enabled,OEN is enabled and write mode */
write_data_coder(group_id);
clock();
     for(i=0;i<80;i++)
      {
      write_ctrl_jpeg(0xbf18);
      fscanf(fdc_tab,"%u",&dc_val[i]);
     temp = (dc_val[i] \&0xff00) >> 8;
      clock();
      write_data_coder(temp);
      temp =dc_val[i] & 0x00ff;
      write_data_coder(temp);
      clock();
      }
 }
 num+=1;
 group_id++;
 dc_table(group_id,num);
}
}
void
ac_table(unsigned group_id,int num)
ł
 int i=64:
 unsigned temp;
 FILE *fac_tab;
 if(num <=1)
  printf(" Do you want to initialize the ac table %d '\n",num);
  if(getch()=='y') {
 printf(" Loading the AC table %d\n",num);
 if (num==0)
 fac_tab=fopen("jpegacey.tab","rb");
 else fac_tab=fopen("jpegacec.tab","rb");
 write_ctrl_jpeg(0xf000); /* OEN is tied low at this stage */
 write_ctrl_jpeg(0xfc18);
 write_ctrl_jpeg(0xbc18); /* '666 is enabled,OEN is enabled and write mode */
 write_data_coder(group_id);
 clock();
      for(i=0;i<1280;i++)
      ł
      write_ctrl_jpeg(0xbf18);
      fscanf(fac_tab,"%u",&ac_val[i]);
      temp = (ac_val[i] \&0xff00) >> 8;
```

```
write_data_coder(temp);
      clock();
      temp =dc_val[i] & 0x00ff;
      write_data_coder(temp);
     clock();
      }
 }
  num+=1;
  group_id++;
  ac_table(group_id,num);
}
}
void
reset(void)
{
outport(BASE + reset45,0x0000);
}
void
init_write(unsigned group_id)
{
int i;
write_ctrl_jpeg(0xf000); /* OEN is tied low at this stage */
write_ctrl_jpeg(0xfc18);
write_ctrl_jpeg(0xbc18); /* '666 is enabled,OEN is enabled and write mode */
write_data_coder(group_id);
clock();
     for(i=0;i<6;i++)
     ł
      write_ctrl_jpeg(0xbf18);
       write_data_coder(dat[i]);
        clock();
     }
}
void
readback_coder(void)
{
unsigned rdval;
int i=0,delay;
write_ctrl_jpeg(0xf000);
write_ctrl_jpeg(0xfc18);
write_ctrl_jpeg(0xfc1c);
/* clock();
clock();
clock();*/
       while(i<6)
        delay=0;
```

```
write_ctrl_jpeg(0xff1c);
            clock();
               rdval = inport(BASE + rdback45);
               rdval = rdval \& 0x00ff;
          while((rdval & 0x00ff)!=dat[i])
           ł
               clock();
                   delay++;
                       rdval = inport(BASE + rdback45);
  printf("rdval = %x and the delay is %d\n",(unsigned)(rdval & 0x00ff),delay);
  i++;
  }
}
void
main(void)
init_write(0x0010);
quant_table_write(0x11,0);
dc_table(0x15,0);
ac_table(0x17,0);
printf(" Do you want to read back\n");
if(getch()=='y')
readback_coder();
else exit(0);
}
/* This program writes to the control register of L64765*/
#include<stdio.h>
#include<stdlib.h>
#include<conio.h>
#include<dos.h>
#define ctrlw 0x2002
#define dataw 0x2004
#define BASE 0x2a0
#define initw 0x2008
```

void
write\_control\_jpeg(unsigned add)
{
outport(BASE+ctrlw,add);
}

93

```
write_data_jpeg(unsigned dat)
ł
  outport(BASE+dataw,dat);
}
unsigned
write_readback_data(void)
{ unsigned ret;
 ret=inport(BASE+initw);
 return(ret);
}
void
main(void)
ł
     unsigned int ret;
     write_control_jpeg(0xf07f);
     write_control_jpeg(0xe07f);
     write_control_jpeg(0xe05f);
     write_data_jpeg(0x08);
     write_control_jpeg(0xe07f);
     write_control_jpeg(0xf17f);
     write_control_jpeg(0xe15f);
     write_data_jpeg(0xe2);
     write_control_jpeg(0xe07f);
     write_control_jpeg(0xf07f);
     write_control_jpeg(0xf07f); /* For reading back the control value */
     write_control_jpeg(0xe07f);
     write_control_jpeg(0xe05f);
     write_data_jpeg(0x08);
     write_control_jpeg(0xf17f); /* For reading back the control value fed in */
     write_control_jpeg(0xf13f);
     ret=write_readback_data()& 0x00ff;
     printf("control value is %02x\n",ret);
     write_control_jpeg(0xf17f);
}
```

/\* This program resets the L64765 \*/

#include <stdio.h>
#include <stdib.h>
#include <conio.h>
#include <conio.h>
#include <dos.h>
#include "init.h"
#define ctrlw 0x2002
#define BASE 0x2a0
#define rw 0x0000
#define cw 0x0006
#define rdback45 0x2007
#define write45 0x2006
#define reset45 0x2000

void
```
unsigned dat[]= \{0x44,0x00,0x00,0x00,0x00,0x01\};
 outport(BASE + cw,0x00);
 outport(BASE + cw,0x04);
 outport(BASE + cw, 0x04);
 outport(BASE + cw, 0x04);
 outport(BASE + cw, 0x04);
 outport(BASE + cw, 0x00);
write_ctrl_jpeg(unsigned dat)
 outport(BASE + ctrlw,dat);
```

```
void
write_data_coder(unsigned dat)
{
 outport(BASE + write45,dat);
}
```

```
void
reset(void)
{
```

void

{

}

{

}

void

clock(void)

write\_ctrl\_jpeg(0xff7f);

## }

```
void
main(void)
{
clrscr();
reset();
clock();
clock();
write_ctrl_jpeg(0xcf7f);
```

## }

/\* This resets the L64745 \*/

#include<stdio.h> #include<conio.h> #include<stdlib.h>

```
#include<dos.h>
#include "init.h"
#define BASE 0x2a0
#define ctrlw 0x2002
#define dataw 0x2004
#define datarb 0x2005
#define ctrlrb 0x2003
#define initw 0x2008
#define rw 0x0000
#define cw 0x0006
#define gw 0x0002
#define bw 0x0004
#define HSYNC 0x08
#define VSYNC 0x02
#define CLK 0x04
unsigned data[48];
void
write_control_rgb(unsigned add)
ł
  outport(BASE + cw,add);
}
void
write_data_rgb(unsigned dat)
ł
 outport(BASE + rw,dat);
}
void
write_control_jpeg(unsigned add)
ł
 outport(BASE + ctrlw,add);
}
void
write_data_jpeg(unsigned dat)
{
 outport(BASE + dataw,dat);
}
void
reset(void)
ł
 write_control_jpeg(0xff7f); /* R[1:0] is set to 11 for reset */
 write_control_jpeg(0xff5f); /* W* is made low */
 write_control_jpeg(0xff7f);
}
```

```
void
main(void)
{
    int i,enc_dec=1;
    unsigned add,ret;
    clrscr();
    reset();
}
```

#include <stdio.h>
#include<stdlib.h>
/\* this program gets the Y U and V components
from the CCIR601 files \*/

```
void main(void)
{
char a[1440];
char input[20];
int ret,i,j;
FILE *fsource, *f1, *f2, *f3;
printf("Separating the YUV components \n");
printf("Input the CCIR601 file\n");
scanf("%s",input);
fsource=fopen(input,"r");
f1=fopen("y1","w");
f2=fopen("u1","w");
f3=fopen("v1","w");
for(i=0;i<480;i++)
 ł
 for(j=0;j<360;j++)
ł
ret=fread(a,1,4,fsource);
if(ret!=4)
 { printf("Error in reading file\n"); exit(1); }
fwrite(&a[1],1,1,f1);
fwrite(&a[3],1,1,f1);
fwrite(&a[0],1,1,f2);
fwrite(&a[2],1,1,f3);
ł
}
 fclose(f3);
 fclose(f2);
 fclose(f3);
 fclose(fsource);
}
```

/\* This program converts SIF files to CCIR601 format \*/

```
#include <stdio.h>
#include <stdio.h>
#include <stdlib.h>
#include <stdlib.h>
/*void CcirName(unsigned char filename[50], int x, int y);
void SifName(unsigned char filename[50], int x);
void
usage(void)
{
printf(" Usage : Sif file ,start frame,end frame, CCIR file"\n);
} */
```

```
/*void ccir_create(void)
ł
 int i,j;
 FILE *fy,*fu,*fv,*fccir;*/
void
main(void)
{
   int i,j,k;
   FILĚ *fsif,*fccir,*fy,*fu,*fv,*fu_temp,*fv_temp;
   unsigned char a[704];
   char pad[16];
   for(i=0;i<16;i++)
   pad[i]=(char)0;
   fsif = fopen("ten.0","rb");
   fy=fopen("y_temp","wb");
fu=fopen("u_temp","wb");
   fv=fopen("v_temp","wb");
printf("Putting Y...\n");
/* Up sampling in X direction */
   for(i=0;i<240;i++)
    Ł
    for(j=0;j<352;j++)
     ł
      fread(a,sizeof(unsigned char),1,fsif);
      fwrite(a,sizeof(unsigned char),1,fy);
      fwrite(a,sizeof(unsigned char),1,fy);
       }
    ł
   fy=fopen("y_in","wb");
fy_temp=fopen("y_temp","rb");
/* Up sampling in y direction */
    for(i=0;i<240;i++)
    {
      fread(a,sizeof(unsigned char),704,fy_temp);
      fwrite(a,sizeof(unsigned char),704,fy);
       fwrite(a,sizeof(unsigned char),704,fy);
    }
```

printf("Putting U....");
/\* Up sampling in x-direction \*/

```
for(i=0;i<120;i++)
   ł
    for(j=0;j<176;j++)
      fread(a,sizeof(unsigned char),1,fsif);
      fwrite(a, sizeof(unsigned char), 1, fu);
      fwrite(a,sizeof(unsigned char),1,fu);
      ł
   }
      fu_temp = fopen("u_temp","rb");
      fu=fopen("u","wb");
/* Up sampling in y-direction */
      for(i=0;i<120;i++)
      fread(a,sizeof(unsigned char),352,fu_temp);
      fwrite(a,sizeof(unsigned char),352,fu);
      fwrite(a,sizeof(unsigned char),352,fu);
      }
    fu=fopen("u_in","wb");
    fu_temp=fopen("u","rb");
   for(i=0;i<240;i++)
      fread(a,sizeof(unsigned char),352,fu_temp);
      fwrite(a, sizeof(unsigned char), 352, fu);
      fwrite(a,sizeof(unsigned char),352,fu);
      }
printf("Putting V....");
   for(i=0;i<120;i++)
   ł
    for(j=0;j<176;j++)
     ł
      fread(a,sizeof(unsigned char),1,fsif);
      fwrite(a,sizeof(unsigned char),1,fv);
      fwrite(a,sizeof(unsigned char),1,fv);
       }
   }
      fv_temp = fopen("v_temp","rb");
      fv=fopen("v","wb");
      for(i=0;i<120;i++)
      fread(a,sizeof(unsigned char),352,fv_temp);
      fwrite(a,sizeof(unsigned char),352,fv);
      fwrite(a,sizeof(unsigned char),352,fv);
       fv=fopen("v_in","wb");
       fv_temp=fopen("v","rb");
```

```
for(i=0;i<240;i++)
{
    fread(a,sizeof(unsigned char),352,fu_temp);
    fwrite(a,sizeof(unsigned char),352,fu);
    fwrite(a,sizeof(unsigned char),352,fu);
    }
}</pre>
```

```
#include <stdio.h>
#include<stdlib.h>
void main(void)
{
unsigned char a[720],b[720],c[720];
int tmpR,tmpG,tmpB,tmpY, tmpU,tmpV;
char tmpr,tmpg,tmpb;
float D = 0.299, E=0.587, F=0.114;
int i,j,ret;
FILE *f4, *f1, *f2, *f3, *f5,*f6;
printf("This program creates the RGB files\n");
/*printf("Input the Y U V files in order n");
scanf("%s\n",y_in);
scanf("%s\n",u_in);
scanf("%s",v_in);
printf("Input the RGB files to be generated in order\n");
scanf("%s\n",r_out);
scanf("%s\n",g_out);
scanf("%s",b_out);
*/
f1=fopen("y1","rb");
f2=fopen("v2","rb");
f3=fopen("u2","rb");
f4=fopen("g1.g","wb");
f5=fopen("g1.b","wb");
f6=fopen("g1.r","wb");
for(i=0;i<480;i++)
 for(j=0;j<720;j++)
ł
ret=fread(a,1,1,f1);
if (ret!=1)
{ printf("Error in reading file\n"); exit(1); }
ret=fread(b,1,1,f2);
if (ret!=1)
 { printf("Error in reading file\n"); exit(1); }
ret=fread(c,1,1,f3);
```

```
if (ret!=1)
{ printf("Error in reading file\n"); exit(1); }
tmpY=a[0];
tmpV = b[0];
tmpV = 128.0;
tmpV = 1.45;
tmpU = c[0];
tmpU -= 128.0;
tmpU = 0.600;
tmpR = tmpY + tmpV;
tmpB = tmpY + tmpU;
tmpG=(tmpY - (D * tmpR)-(F * tmpB))/E;
if (tmpR < 0) tmpR = 0;
if(tmpR > 255) tmpR = 255;
if(tmpG > 255)tmpG = 255;
if (tmpG < 0) tmpG = 0;
if (tmpB < 0) tmpB = 0;
if(tmpB > 255) tmpB = 255;
tmpr = (char) tmpR;
tmpg=(char)tmpG;
tmpb=(char)tmpB;
fwrite(&tmpg,1,1,f4);
fwrite(&tmpb,1,1,f5);
fwrite(&tmpr,1,1,f6);
fclose(f4);
fclose(f5);
fclose(f6);
fclose(f1);
fclose(f2);
fclose(f3);
}
#include <stdio.h>
#include <stdlib.h>
#include "tgarhb.h"
TGAHeader th =\{0,0,2,0,0,0,0,0,512,480,16,32\};
void
main(void)
{ int i,j,ret;
   char a,b,c,targa[20];
   FILE *fr,*fg,*fb,*ftarga;
/* printf(" What is the R file\n");
   scanf("%s",r_in);
   printf("What is the G file \n");
   scanf("%s",g_in);
    printf("Input the B file \n");
    scanf("%s",b_in); */
    printf("What is the output targa file\n");
```

```
if( (fr = fopen("g2.r", "rb")) == NULL)
   printf("Unable to open the R file\n"):
   exit(1);
if (fg=fopen("g2.g","rb"))==NULL)
   printf("Unable to open the G file\n");
   exit(1);
if ((fb=fopen("g2.b","rb"))==NULL)
 {
  printf("Unable to open the B file n");
  exit(1);
if( (ftarga = fopen (targa, "wb"))==NULL)
 ł
   printf("Unable to open the targa file\n");
    exit(1);
 }
fwrite((char *)&th.idLength ,sizeof(UINT8),1,ftarga);
fwrite((char *)&th.mapType ,sizeof(UINT8),1,ftarga);
fwrite((char *)&th.imageType ,sizeof(UINT8),1,ftarga);
fwrite((char *)&th.mapOrigin,sizeof(UINT16),1,ftarga);
fwrite((char *)&th.mapLength,sizeof(UINT16),1,ftarga);
fwrite((char *)&th.mapWidth ,sizeof(UINT8),1,ftarga);
fwrite((char *)&th.xOrigin ,sizeof(UINT16),1,ftarga);
fwrite((char *)&th.yOrigin ,sizeof(UINT16),1,ftarga);
fwrite((char*)&th.imageWidth,sizeof(UINT16),1,ftarga);
fwrite((char*)&th.imageHeight,sizeof(UINT16),1,ftarga);
fwrite((char *)&th.pixelDepth,sizeof(UINT8),1,ftarga);
fwrite((char *)&th.imageDesc,sizeof(UINT8),1,ftarga);
/* fwrite((char *)&th.idString,sizeof(char),1,ftarga); */
for(i=0;i<480;i++)
   for(j=0;j<512;j++)
 ł
   ret=fread(\&a,1,1,fr);
   if(ret!=1)
   printf("Error in reading the file\n");exit(1); }
   ret=fread(\&b,1,1,fg);
      if(ret!=1){
   printf("Error in reading the file\n");exit(1); }
   ret=fread(\&c,1,1,fb);
      if(ret!=1)
   printf("Error in reading the file\n");exit(1); }
   fwrite(&c,1,1,ftarga);
   fwrite(&b,1,1,ftarga);
   fwrite(&a,1,1,ftarga);
```

```
}
}
}
/*
   PROGRAM snr.c
                              */
/* This program computes the average SNR
 The user has to enter the following:
 - original file name
 - reconstructed file name
  - appropriate delay
 - frame size */
#define BORLANDC 1
#include <stdio.h>
#include <fcntl.h>
#if BORLANDC==1
#include <string.h>
#include <stdlib.h>
#else
#include <strings.h>
#endif
#include <math.h>
#define LS 160
#define nxt_line \n'
void inputname(char *filename,int i);
void outputname(char *filename,int i);
double unsigned_int_dbl(int vec2);
double vvdot(double vec1,double vec2);
double vvdiff(double vec1,double vec2);
unsigned char filename[50];
double unsigned_int_dbl(int vec2)
{ double vec1;
    vec1 = (double) vec2;
  return(vec1);
}
double vvdiff(double vec1,double vec2)
{
  double vec3;
    vec3 = vec1 - vec2;
  return(vec3);
 }
void usage(void)
 {
```

```
printf("\nUsage :snr \\\n");
                       printf("\ninput data are short !!! \\\n");
                       printf("\t original image file\\\n");
                       printf("\t start frame\\n");
                       printf("\t length\\\n");
                       printf('\t output file\\\n");
 }
double vvdot(double vec1,double vec2)
         double result;
         result = 0.0;
                result += vec1*vec2;
         return(result);
}
void main(int argc,char** argv)
infile[100],outfile[100];
/* option flags
                    */
int length, start_s;
FILE *infile1,*infile2,*fsnr; /* original and reconstructed files */
long countl; /* number of samples read from original image */
long count2; /* number of samples read from reconstructed image */
long m; /* number of samples read from reconstructed image every iteration */
int nframe, nframeb; /* # of frame */
int i; /*iteration variable*/
double x; /* original image */
short int xx; /* original image */
double y; /* reconstructed image */
short int yy; /* reconstructed image */
double e: /* error */
double sigpow; /* energy of the original image */
double sqrerr; /* energy of the error */
double totpow; /*total average power of input */
double toterr; /*total average power of error */
float snr_val; /* Signal to Noise Ratio */
fsnr=fopen("snrout","wb");
/* parse and read command line arguments */
        if(argc!=5){
                usage();
                exit(1);
         }
         strcpy (infile,argv[1]);
         strcpy(outfile,argv[4]);
         start s = atoi(argv[2]);
         length = atoi(argv[3]);
```

```
length = atoi(argv[3]);
  for(i=start_s;i<length;i++)
{
  strcpy(filename,infile);
  inputname(filename,start_s + i);
  printf("file : %s \n",filename);
  infile1 = fopen(filename,"rb");
 strcpy(filename,outfile);
 outputname(filename, start_s + i);
  if((infile2 = fopen(filename,"rb"))==NULL)
{ printf( "Unable to open the out put file \mn");
  exit(0);
}
   printf("file : %s\n",filename);
  printf("\n\n");
  printf("\n\n");
  /* positioning the reconstructed speech file pointer accordingly */
  /* intialization */
   totpow=toterr=0.0;
   count1=count2=nframe=nframeb=0;
  /* read first frame samples from each file */
   fread(&xx,sizeof(int),1,infile1);
   m=fread(&yy,sizeof(int),1,infile2);
   x=unsigned_int_dbl(xx) ;
   y=unsigned_int_dbl(yy);
   while (m>0)
   {
      nframe++;
      count1=count1+1;
      count2=count2+1;
      sigpow=vvdot(x,x);
      sigpow=sigpow/m;
      if (sigpow > 0) {
      totpow +=sigpow;
      e=vvdiff(x,y);
      sqrerr=vvdot(e,e);
      sqrerr=sqrerr/m;
 /* printf("error power at frame%d=%f\n",nframe,sqrerr);*/
      toterr +=sqrerr;
      }
```

```
else
         ł
         nframeb++;
       fread(&xx,sizeof(short int),1,infile1);
       m=fread(&yy,sizeof(short int),1,infile2);
       x=unsigned_int_dbl(xx);
       y=unsigned_int_dbl(yy);
   }
   fclose(infile1);
   fclose(infile2);
   /*totpow =totpow/(double) (nframe-nframeb);
   toterr =toterr/(double) (nframe-nframeb);*/
   if (toterr == 0.0) printf(" Infinite SNR n");
     else {
         snr_val=10*log10(totpow/toterr);
         printf("SNR = %lf\n",snr_val);
         fprintf(fsnr,"%d) %lf\n",i,snr_val);
      fclose(fsnr);
      fclose(infile1);
      fclose(infile2);
void inputname(char *filename,int i)
   unsigned char tmpstring[20];
   strcat(filename,".");
   if(i < 10)
   sprintf(tmpstring,"%1d",i);
 else if(i < 100 \&\& i > = 10)
   sprintf(tmpstring,"%2d",i);
 else if(i \ge 100)
   sprintf(tmpstring,"%3d",i);
 strcat(filename,tmpstring);
void outputname(char *filename,int i)
   {
      unsigned char tmpstring[20];
      strcat(filename,".");
      if (i < 10)
   sprintf(tmpstring,"%ld",i);
   else if(i \ge 10 \&\& i < 100)
         sprintf(tmpstring,"%2d",i);
   else if(i \ge 100)
   sprintf(tmpstring,"%3d",i);
      strcat (filename,tmpstring);
   }
```

}

}

## REFERENCES

- 1. Dr Rashid Ansari, Fure Ching Jeng, Andria Wong. *MPEG Simulation Manual*, 1990.
- 2. William Pennebaker, Joan L. Michell. JPEG Still Image Data Compression Standard, Van Nostrand Reinhold, NY, 1993.
- 3. LSI LOGIC: LSI Chipset manual for JPEG algorithm, 1990.
- 4. F.Rocca and S. Zanoleth. *Bandwidth reduction via movement compensation on a model of a random video process*, IEEE trans. Comm, vol COM-20, pp 960-965, Oct 1972.
- 5. JPEG Committee. JPEG Draft .1989.
- 6. A.K. Jain. Fundamentals of Digital Image Processing, Englewood Cliffs, NJ: Prentice Hall, 1989.
- 7. N.Ahmed, T.Natarajan, and K.R Rao. *Discrete Cosine Transform*. IEEE Transactions, Computers, C-2390-3 (Jan 1974).
- 8. P.Elias and N.Abramson. *Information theory and Coding*. New York: McGraw Hill Book Co, 1963.
- 9. M.Rabbani and P.W Jones. *Digital Image Compression Techniques*. Vol TT7. pp 1-221, Bellingham, WA:SPIE.
- R.B Arps, T.K Truong, D.J Lu, R.C Pasco, and T.D Friedman. A multi purpose VLSI chip for adaptive data compression techniques. IBM J.Res. Develop. 32(6):775-95 (Nov.1988).
- 11. J.G Cleary, I.H Witten, and R.M. Neal. *Arithmetic Coding for Data Compression*. Communic of the ACM. 30(6):520-40 (Jun 1987).
- 12. A.LigtenBerg, R.H Wright, and J.H.O'Neill. A VLSI Ortogonal Transform for Real time Image Compression. Visual Communication & Image Process. (Oct 1987).
- 13. B.D. Tseng and W.C Miller. On Computing the Discrete Cosine Transforms. IEEE Trans. on Computers. C27(10):1966-8.
- 14. D.A Huffman. A method for the construction of Minimum-Redundancy Codes. Proc. IRE. 40(9):1098-101 (Sept 1952).
- 15. D.S Thornton and K.L Anderson. A BNF for the JPEG Image Compression Header Structure. Proceedings Electronic Imaging '90 East, pp96-99. Boston, MA (Oct 29- Nov 1,1990).
- 16. G.G Langdon and J.J Rissanen . Compression of Black and White Images with Arithmetic coding. IEEE trans. on Commun. 29(6):858-67 (Jun. 1981).

- 17. C.B Jones. An Efficient Coding System for Long Source Sequences. IEEE Trans Info Theory. IT-27(3):280-91 (May 1991).
- 18. JJ Rissanen and G.G Langdon. Arithmetic Coding. IBM J. Res. Develop. 23(2):149-62 (Mar. 1979).
- 19. J.G Cleary, I.H Witten, and R.M. Neal. Arithmetic Coding for Data Compression. Commu. of the ACM. 30(6):520-40 (Jun.1987).